

Network Centrality and Pension Fund Performance

Alberto G. Rossi[†]

David Blake[‡]

Allan Timmermann[§]

Ian Tonks[¶]

Russ Wermers[∥]

Abstract

We analyze the relation between the location of a pension fund in its network and the investment performance, risk taking, and flows of the fund. Our approach analyzes the centrality of the fund's management company by examining the number of connections it has with other management companies through their commonality in managing for the same fund sponsors or through the same fund consultants. Network centrality is found to be positively associated with risk-adjusted return performance and growth in assets under management, after controlling for size and past performance, for domestic asset classes; however, we do not find this relation for foreign equity holdings. These findings indicate that local information advantages, which are much stronger among managers holding locally based stocks, exhibit positive externalities among connected managers. Of particular note is that we do not find that the centrality of a manager within one asset class (e.g., domestic bonds) helps the performance of the manager in another asset class (e.g., domestic equity), further indicating that our network analysis uncovers information diffusion effects. Network connections established through consultants are found to be particularly significant in explaining performance and fund flows, consistent with consultants acting as an important information conduit through which managers learn about each other's actions. Moreover, the importance of network centrality is strongest for larger funds, controlling for any economic scale effects. Better connected funds are also better able to attract higher net inflows for a given level of past return performance. Finally, more centrally placed fund managers are less likely to be fired after spells of low performance. Our results indicate that networks in asset management are one key source of the dissemination of private information about security values.

[†]Smith School of Business, University of Maryland-College Park; Email: arossi@rhsmith.umd.edu.

[‡]Cass Business School, City University of London; Email: d.blake@city.ac.uk.

[§]Rady School of Management, University of California at San Diego; Email: atimmermann@ucsd.edu.

[¶]University of Bath School of Management; Email: it237@management.bath.ac.uk.

Smith School of Business, University of Maryland-College Park; Email: rwermers@rhsmith.umd.edu.

1 Introduction

The efficient dissemination and processing of information is central to the success of professional money management. The best money managers must both be well-connected to sources of private information on stock values, and be able to quickly interpret and apply the information that they receive from their connections to their portfolio choices. That is, access to information, including information derived from knowledge of competitors' investment strategies, is a likely mechanism though which funds may achieve abnormal performance.

While the econometrician cannot observe the private information that flows to individual fund managers-let alone how managers process such information-the centrality of managers in a network of institutional investors may plausibly be expected to be associated with the breadth of a manager's information on investment strategies and opportunities. Even if better connections do not translate into specific information about "stock picks," better connected managers are more likely to be aware of fund flows into specific asset classes and individual stocks, and, so, can strategically position themselves to take advantage of this, reducing the liquidity effects of redemptions or withdrawals.

In this paper, we use data on a large set of UK pension fund accounts over the period 1984-2004 to ask whether network centrality explains managers' investment performance, risk-taking behavior, and flows. The dataset, also described in Blake et al. (2013), allows us to compute network connections in a unique manner.¹ Moreover, we have data on investment performance and network connections within three asset classes–UK Equity, UK Bonds, and International Equity–that together account for more than 85% of the funds' asset holdings. This allows us to compare and contrast findings across asset classes for which we would expect network effects to be very different–for example, we would not expect network connections within the UK to be as important to performance in foreign markets as in domestic markets, as access to information about one's competitor should yield the biggest benefits in local markets. As such, our large cross-section and time-series of managers, consultants, and sponsors provides several dimensions through which we can identify the impact of networks on information diffusion (since networks can respond to, and not simply predict changes in the information setting).

 $^{^{1}}$ Blake et al. (2013) focus on the effect of the recent decentralization of pension fund management on investment performance, while we focus on the network connections between fund managers, fund sponsors, and investment consultants.

Network connections in our dataset arise from two separate sources. First, many of the pension funds use multiple managers (each managing a separate account for the sponsor), and such overlaps, in turn, create a network of connections across fund-manager pairings. For example, UBS might manage a portion of the accounts of each of dozens of pension fund clients, many of whom have also hired Merrill Lynch. This creates a strong "connection" between UBS and Merrill Lynch. As a second source of network connections, pension funds hire consultants to provide advice on which managers to hire, thus creating a network between, say, UBS and Merrill Lynch, through having a common consultant in their interactions with several fund sponsors. Such relationships help reduce the search costs incurred by the pension fund trustees. In summary, individual pension fund accounts can be connected by their sharing of the same consultant and/or the same manager.

Our dataset allows us, uniquely, to analyze not only the relation between network centrality and key variables such as investment performance, risk-taking behavior, and fund flows, but also which type of network connections-through shared managers or through shared consultants-matter most.² This is important since little is known about the specific channels through which information flows in the fund management business. One possibility is that information flows through the "overlap" of individual managers. By sharing the same large institutional clients, managers can expect to learn valuable information about their competitors' investment strategies, industry preferences, and resulting investment performance. Alternatively, consultants play a key role in advising fund trustees in selecting, reappointing and firing fund managers. As monitors of the fund management industry, consultants are a natural conduit through which information flows on individual fund manager performance and investment style. In fact, the entire justification for the existence of consultants arises from their ability to identify successful fund-manager matches.

Our empirical analysis uncovers a number of novel findings. First, we find that fund-managers that are better connected (or more central in the network) tend to have higher risk-adjusted performance in both UK equities and UK bonds. This result is not explained simply by more central managers being bigger. In fact, while there is a positive correlation between manager size and centrality, size is negatively related (consistent with Chen, Hong, Huang, and Kubik, 2004), while centrality is positively

 $^{^{2}}$ To capture network centrality, our analysis focuses on degree centrality, which measures the probability that a node (fund-manager pairing) immediately captures newly released information through its direct contacts with other nodes (fund-managers).

related to risk-adjusted performance. We also find an interesting interaction effect between fundmanager size and network centrality which suggests that large funds have a particular advantage when it comes to benefitting from network centrality. While large funds tend to underperform, large funds that are also well connected manage to reduce the negative effect of fund size. In contrast, we find little evidence that centrality within our network of UK managers affects the risk-adjusted performance in international equities, consistent with more 'localized' benefits associated with network centrality.

Second, across all three asset classes, we find that network connections have a large and significantly positive effect on fund flows, again after controlling for size. In contrast, fund size has a negative effect on fund flows, whereas past returns fail to significantly impact fund flows. This suggests that, controlling for size and past returns, the more central a manager is, the greater the expected inflows tends to be. Moreover, interaction terms between centrality and past returns are strongly positively associated with fund flows, even though the separate effect of past return is insignificant. This suggests that being centrally placed in the network allows a fund to positively exploit its past performance record to grow its assets under management. Conversely, large and poorly connected fund managers feel the full negative impact of their size and experience smaller inflows.

Network centrality may influence funds' risk-taking behavior through managers' (or consultants') incentives. Specifically, network centrality could affect managers' risk-taking behavior through its effect on fund flows and, thus, via assets under management, their remuneration. Our third main empirical results establishes evidence that better connected funds tend to reduce their idiosyncratic risk taking within domestic equities and bonds relative to less central funds. This effect is attenuated by fund size: an interaction term between fund-manager size and centrality has a strongly negative effect on idiosyncratic risk taking, suggesting that large, centrally placed managers hold portfolios with less idiosyncratic risk. For international equities we find the opposite effect: better-connected managers increase their idiosyncratic risk levels compared with less central managers, and the effect is particularly strong for the largest managers through the interaction term.³

Fourth, network centrality may also influence managers' or consultants' behavior through the

 $^{^{3}}$ Previous studies (e.g., Kang and Stulz (1997)) have found that foreign investors act as if they are at an informational disadvantage abroad. Our finding is consistent with the larger, more central managers–who are likely to have international offices and networks of their own–being less cautious when investing abroad than the smaller managers.

probability that they are fired by their pension fund clients. To explore if this is the case, we estimate semi-parametric Cox regressions that relate managers' or consultants' hazard rate (the probability that they are fired next period) to the tenure of the fund-manager relation, manager size, past return performance and network centrality. We find strong evidence that more central managers face a significantly reduced probability of being fired, after controlling for size and past return performance. This reveals another incentive for managers to establish network connections: it makes them more immune towards being dismissed by their institutional clients.

Fifth, to the best of our knowledge, no previous academic research paper has analyzed the role of consultants in the investment decisions of institutional clients and estimated its effect on investment outcomes. Consultants introduce an extra layer of agency with their own incentives and are subject to the same process of hiring and firing as the managers whom they monitor. For many of our models we find that the consultant-based networks add information over and above the information contained in the networks established solely through manager connections. Thus consultants' role in the investment process is not subsumed by that of the managers.

Our results reveal rich dynamic interactions between fund-manager centrality, risk-adjusted performance, fund flows and thus, ultimately, fund size. Our final empirical contribution addresses whether size drives centrality or centrality drives size by conducting panel Granger causality tests. These suggest that network centrality Granger causes size, while size does not Granger cause network centrality. In particular, central fund managers tend to grow larger faster than more peripheral fund managers, after controlling for past size.

Our results suggest a new mechanism for why some fund managers are successful, while others are not. Being centrally located in the fund-manager network appears to foster better risk-adjusted investment performance, higher inflows and an ability to reduce the negative impact of size that affects most funds as they grow large.

The plan of the paper is as follows. Section 2 introduces our data and presents evidence on both manager- and consultant-based networks at fixed points in time as well as throughout our sample. Section 3 explores the relation between risk-adjusted return performance and network centrality, while Section 4 considers the dynamic relation between fund flows and network centrality by estimating models that relate fund flows to past flows, size, return performance, and network centrality. Section 5 considers how funds' risk-taking behavior is linked to their network centrality and analyzes if centrality affects managers' incentives through their risk of being fired. Section 6 presents results from a Granger causality analysis of the size-centrality relation, while Section 7 concludes.

2 Data and network centrality measure

This section introduces the data on UK pension funds used in our study and depicts the networks established between funds, managers and consultants for three asset classes–UK equity, UK bonds and International equity–that are central to the pension funds' portfolio holdings. Next, we describe how we construct the centrality measure used in our analysis (degree centrality) and provide insights into its characteristics, its evolution over time, and its correlation with other variables in our dataset.

2.1 Data

Our dataset comprises quarterly returns and asset holdings of 2,385 occupational defined benefit pension plans between March, 1984 and March 2004. The data, which was provided by BNY Mellon Asset Servicing, has information on seven asset classes, but we concentrate on the three biggest ones–UK Equity, International Equity, and UK Bonds–which together comprise around 85% of asset holdings by market value throughout the sample.⁴ For each fund, and within each asset class, we know the identity of the fund manager–or managers in cases with multiple managers–at each point in time. This is important since it is common, especially for large funds, to hire different managers for different asset classes; moreover, funds commonly employ two or more specialized managers within the same asset class, e.g., a large and a small cap equity manager.

Such overlaps create the possibility of network effects as fund sponsors will want to coordinate the investment decisions across different managers so as to minimize the inefficiency loss associated with decentralized decision making (e.g., Sharpe (1981), van Binsbergen et al. (2008), and Blake et al. (2013)). Funds may also indirectly reveal information about other managers' investment strategies by setting up competition among managers, ensuring that the best-performing managers within a partic-

⁴The other asset classes are cash, International bonds, index-linked bonds, and property.

ular asset class see their assets under management increased at the expense of the worse-performing competitors.

2.1.1 Managers

Table 1 shows the number of fund-manager pairings—the unit of observation for much of our analysis at three points in time (1984, 1994, and 2004).⁵ Within UK equities the number of fund-manager pairings starts at 1204, increases to 1420, only to decline to 1053 at the end of the sample. A similar pattern emerges for international equities, where the count of fund-manager pairings is 1135, 1354 and 956 in 1984, 1994 and 2004, respectively. UK bonds behave slightly differently as the number of fund-manager pairings decreases from 1165 to 745 during the decade 1984-1994, then increases to 817 in 2004. The increase in the number of fund-manager pairings for UK bonds reflects the increased prominence of this asset class towards the end of sample as many defined benefit pension schemes switched their assets towards domestic bonds.

Table 1 also reports the number of funds in the dataset at the same three points in our sample. Between 1984 and 1994 the number of UK equity and international equity funds increased slightly, while conversely it decreased for UK bonds. The number of funds then decreased for all asset classes between 1994 and 2004. Comparing the number of fund-manager pairings to the number of funds, it is evident that over our 20-year sample a large number of funds moved from being single-managed to being multi-managed - a change in paradigm analyzed in details in Blake et al. (2013). For example, the average number of UK equity managers per fund went from 1.26 in 1984 to 1.67 in 2004.⁶

The remaining columns of Table 1 present the number of managers as well as summary statistics for the number of connections per manager. The number of UK equities managers in our sample declined from 113 in 1984 to 82 in 2004. An even sharper decline is observed for UK bond managers (from 109 to 61), while the decline was more modest for international equities (from 108 to 89). At the same time, the number of network connections per manager increased over time, indicating that the pension fund management industry became more inter-connected over the sample. For example, the proportion of managers with more than 20 network connections increased from 7% in 1984 to 12%

⁵Each time a manager manages a portion of a fund's assets, a separate account is set up whose assets and return performance are tracked through time.

 $^{^{6}}$ Namely, from 1204/955 to 1053/630.

of the managers in 2004. Similar patterns are seen for UK bonds and international equities.

2.1.2 Consultants

The pension funds in our sample are advised by consultants who assist in the appointment of fund managers and the choice of investment mandates. A total of 12 different consultants performed these services over our sample period. For each consultant Figure 1 shows time-series plots of the number of clients in UK equities (Panel A), UK bonds (Panel B) and international equities (Panel C). The number of clients advised by the individual consultants appears to follow very similar patterns across the three asset classes, indicating that the consultants did not specialize in specific asset classes. This is consistent with the view that consultants provide 'full service'' advice to their pension fund clients, as is required when selecting managers with multi-asset class or so-called balanced investment mandates.

Figure 2 shows the proportion of total assets in UK equities, UK bonds and international equities advised by each of the consultants, i.e., their market shares by asset value. This figure suggests that the market for consultants was dominated by four large firms whose combined market size did not change much over the sample period.⁷

2.2 Networks In different asset classes

A network is characterized by its nodes (agents) and edges (connections). To construct networks we include all agents that are present at a given point in time. This allows us to construct time series of network connections. Consultants act as interlocutors in the network, so we consider separately network connections established either through consultants or managers as well as networks in which connections are established exclusively through managers (managers only) or exclusively through consultants (consultants only). In the manager-only case managers are connected only if they comanage the assets for one or more pension funds either as joint balanced asset managers or as specialist managers of separate asset classes. In the consultant-only case, two managers can be connected if any of the clients whose portfolios they manage share the same consultant (e.g., pension fund P uses

 $^{^{7}}$ In 1998, consultant no. 11 merged with consultant no. 2; the merger is evident in all panels in Figure 1. However, the effect of this merger on (consulting) industry concentration was not very pronounced. In fact, while Figure 1 shows that consultant no. 11 was responsible for advising a relatively large number of funds, Figure 2 shows that these funds were typically of small size.

Manager A, while pension fund Q uses Manager B; both pension funds use the same Consultant C); in this case, the two managers are connected through the common consultant although they do not share the same (pension fund) client.

Figure 3 shows network connections in UK equities at three points in time during our sample, namely 1984, 1994, and 2004. Nodes shown as red circles represent individual managers, while the black diamonds in the horizontal row represent the twelve consultants. Next to each node is shown the code of the manager or consultant. This is specific to individual managers and consultants and remains constant throughout the sample. Managers whose nodes are shown above the consultants are only connected through the consultants, while the managers whose nodes fall below the consultants are connected with at least one other manager.

Network connections can be established either through manager or through consultants. Specifically, blue lines in Figure 3 track network connections between managers (established through managers' sharing of the same pension fund client) while green lines track connections between consultants and managers (established through consultants advising the same pension funds). While the two are of course related–consultants are more likely to favor certain managers over others–the figure nevertheless shows how we can separately measure the two types of connections.

Figure 3 reflects that there are far more managers than consultants–although consultants generally have many more connections than the typical manager. In 1984, six consultants had far more connections than the remaining ones; In 1994, nine of the 12 consultants have multiple connections, while in 2004 this number decreased to seven consultants, showing the consolidation that took place in the consulting industry over the 20-year sample period. Thus, the number of network connections among managers tells a similar story to that provided by market shares plotted in Figure 2.

Figures 4 and 5 display the corresponding networks for UK Bonds and International Equity. UK bond managers in particular had far fewer network connections than UK Equity managers with International Equity managers falling in the middle. Moreover, the networks also appear to have evolved very differently through time. For example, while the UK Equity network became more dense between 1984 and 2004, the opposite appears to have happened for UK bonds, where the number of connections actually declined over the sample.

Figure 6 shows a three-dimensional plot of network connections at the end of the sample (2004) for the three asset classes. This figure contains the same information as the bottom plots in Figures 3-5 but presents it in an alternative way. Green balls represent fund managers while the yellow balls represent consultants; the size of each ball is proportional to its centrality in the network.⁸ The plots indicate that the UK equities network is populated by both highly connected managers and consultants that are very central to the network as well as a number of peripheral managers. The network in international equities is more dispersed in that the number of managers in the very center of the network is smaller than for UK equity. Finally, the network in UK bonds does not seem to be populated by managers in the very center as the managers are more distant from each other.

2.3 Measuring network centrality

We use degree centrality as our measure of network centrality. Degree centrality measures the number of neighbors a node has relative to the total number of nodes. For a specific network node this measure can be interpreted as the immediate probability that the node "catches" information flowing through the network. A node can be a manager or a consultant because we can think of either as conduits for information flows.

In our context, if a particular manager (or consultant) is in possession of some information, the probability that this information gets transferred to another manager (consultant) next period is a function of the number of contacts (nodes) the manager (consultant) is adjacent to. The degree measure focuses on short-term information spillover since there is no notion of connectedness through a chain of nodes—only direct connections count for this measure. As such, the measure captures more localized information flows; these are likely to be more strongly correlated with investment performance than more global measures of centrality such as closeness centrality.⁹

⁸The layout of the plot is obtained by using the Fruchterman-Reingold 3D force-directed layout algorithm with a factor of 3. The idea behind the Fruchterman-Reingold 3D algorithm is to represent the nodes as steel rings and the edges as springs between them and consider attractive and repulsive forces between them. The attractive force is analogous to the spring force and the repulsive force is analogous to the electrical force. The algorithm minimizes the energy of the system by moving the nodes and changing the forces between them.

⁹Other measures of network centrality such as betweenness centrality and prestige centrality have also been proposed. However, these are either not appropriate for measuring a node's importance for the flow of information (betweenness) or focus on longer-term effects of information flow (prestige) which are likely to be less relevant to financial networks such as those studied here in which information can be expected to flow fast.

Formally, the degree centrality " DE_{jt} " of node j at time t is defined as:

$$DE_{jt} = \frac{d_{jt}}{N_t - 1},\tag{1}$$

where d_{jt} is the number of connected neighbors for node j at time t and N_t is the total number of nodes in the network at time t.

Figure 7 provides a simple example showing how closeness and degree centrality are computed in a simple network with 7 nodes. Node 1 has one connected neighbor (node 2), leading to a degree centrality measure (using equation 1) of $\frac{1}{7-1} = 0.17$. The same degree centrality characterizes nodes 2, 6 and 7. Nodes 3 and 5 have degree centrality of $\frac{3}{7-1} = 0.5$ because they are each connected to three nodes. Finally, node 4 is connected to 2 nodes and so has a degree centrality of $\frac{2}{7-1} = 0.33$.

2.4 Evolution in Networks

Figures 3-5 suggest that the number of network connections within each of the three asset classes has changed substantially over time. To get a better sense of how the "average" centrality measure evolved during our sample, we next study the time-series of average degree centrality. Specifically, we first standardize each centrality measure " CM_t " by subtracting its time-series average and standard deviation over the full sample so as to create a measure with mean zero and unit variance. Specifically, for each asset class the standardized centrality measure is constructed as follows:

$$\widetilde{CM}_t = \frac{CM_t - MEAN(CM_t)}{STDEV(CM_t)},\tag{2}$$

where $CM_t = N_t^{-1} \sum_{j=1}^{N_t} CM_{jt}$ is the cross-sectional average centrality measure at time t, and $Mean(CM_t)$ and $STDEV(CM_t)$ are time-series averages of CM_t computed over the sample 1984-2004.

Figure 8 plots time series of the normalized centrality measures over our sample. We show the total centrality measure based on network connections established across either managers or consultants ("total", shown as a black line) as well as the separate measures established across managers only (blue dotted line) and consultants only (red dashed line).

For UK Equity (upper left corner), we see a distinct upward trend in both the total and manageronly centrality measures after 1990, while the consultant-only centrality measure displays more evidence of mean reversion, particularly during 1998-2002. Almost identical patterns are seen for International Equities, while all three centrality measures trend upwards for UK bonds (upper right corner).

Arguably, information across managers is most likely to flow within the same asset class. For example, manager A is likely to learn more from manager B's actions if they both advise the same client on UK equities than if manager A is a domestic equity manager and manager B is a bond manager. Figures 3-6 therefore show results based on the individual asset classes. However, we can also construct networks that allow for connections across asset classes (e.g., if a UK Equity manager and a UK Bond manager share the same pension fund client). The bottom right window in Figure 8 shows that such network connections trend upwards for the total and manager-only measures but mean revert for the consultant-only measure.

These plots indicate that the consultant-only and manager-only measures, though clearly sharing a common component, display quite different behavior and so are likely to capture different information. To explore the relation between the centrality measures and other variables, Table 2 uses correlations to summarize the relation between the overall ("total") network centrality measure established across either managers or consultants (NET) versus the two separate centrality measures established across managers-only (NET_M) or consultants-only (NET_C). Across all three asset classes we find a strongly positive (0.78-0.81) correlation between the manager-only and consultant-only network centrality measures. As we shall see in subsequent analysis, the far from perfect correlation between the manager-based and consultant-based centrality measures allow us to identify whether changes in manager behavior is induced through manager/consultant connections or through both.

In turn, the manager/consultant centrality measures are nearly uncorrelated with the fund-manager size but have a positive correlation (0.48-0.56) with manager size. We would expect larger managers to have more network connections, but the results here suggest that manager size only accounts for a modest proportion of the variation in network centrality, raising the prospects that we can identify the separate effect of network centrality and size on managers' investment performance. We next address

this question.

3 Return performance and network centrality

This section relates the network centrality measures introduced in the previous section to risk-adjusted return performance in the three asset classes (UK equities, UK bonds and International equities) that we study. We first explain how we construct the dependent variable (risk-adjusted returns) for each asset class and then present results from panel regressions that use centrality as a covariate, while controlling for fund/manager size and other variables. Finally, we also present non-parametric tests based on portfolio sorts using centrality or centrality and size as sorting variables.

3.1 Risk-adjusted return regressions

To explore the relation between risk-adjusted returns and network centrality, we first construct an estimate of risk-adjusted returns using a procedure similar to that in Blake et al. (2013).¹⁰ Specifically, for each fund-manager pairing, we compute quarterly UK equity returns net of a three-month risk-free rate, r_{ijt} . Here, the subscript 'i' refers to the fund, while 'j' refers to the manager and 't' refers to the time period. We next regress this on an intercept, excess returns on the UK stock market portfolio, $r_{mkt.t}$, returns on a size factor, SMB_t , a value-growth factor, HML_t and a momentum factor, MOM_t :

$$r_{ijt} = \alpha_{ij} + \beta_{1ij}r_{mkt,t} + \beta_{2ij}SMB_t + \beta_{3ij}HML_t + \beta_{4ij}MOM_t + \varepsilon_{ijt}.$$
(3)

For UK bonds we estimate a two-factor model using an intercept and excess returns on the FTSE All-Gilts Total Return Index (GOVB) and UK government consol bonds (*CONS*) as regressors:

$$r_{ijt} = \alpha_{ij} + \beta_{1ij} GOVB_t + \beta_{2ij} CONS_t + \varepsilon_{ijt}.$$
(4)

Finally, for international equities, we use a four-factor model that includes an intercept, sterlingdenominated excess returns on the MSCI North American (NA) and Europe Australasia Far Eastern

¹⁰Since we are not interested in studying market timing skills, and since Blake et al. (2013) only found weak evidence that managers have market timing skills, we omit the market timing terms from the performance regressions here.

ex-U.K. (EAFEX) Total Return Indices as well as global size (SMB) and value-growth (HML) factors:¹¹

$$r_{ijt} = \alpha_{ij} + \beta_{1ij}NA_t + \beta_{2ij}EAFEX_t + \beta_{3ij}SMB_t + \beta_{4ij}HML_t + \varepsilon_{ijt}.$$
(5)

To estimate these models, we drop from the dataset fund-manager pairings that survive less than 12 observations. Using the resulting estimates, for each fund-manager and each quarter, we compute the associated risk-adjusted returns, $\hat{r}_{ijt}^{adj} = \hat{\alpha}_{ij} + \hat{\varepsilon}_{ijt}$.

3.2 Risk-adjusted returns and network centrality

We next perform panel regressions using the risk-adjusted returns, computed as described above, as our dependent variable. To control for size effects in the risk-adjusted return regressions, we include two terms. First, we compute the size for each fund-manager pairing, labeled $size_{ijt}$ and measured as the market value of the assets managed by manager j for fund i at the beginning of quarter t. Second, for each manager we compute the assets under management in UK equities across all funds managed at time t, labeled $Msize_{jt} = \sum_{i=1}^{N_{it}} size_{ijt}$. Each quarter we convert these to relative size measures by taking the log of the size variable divided by its cross-sectional average, e.g., $\log(Msize_{jt}/Msize_t)$, where $\overline{Msize_t} = N_{jt}^{-1} \sum_{j=1}^{N_{jt}} Msize_{jt}$ is the cross-sectional average manager size at time t.

Using our estimates of risk-adjusted performance from equations (3)-(5), we perform panel regressions that allow for both fund-manager and time fixed effects, control for fund-manager and manager size effects and use standard errors that are clustered at the fund-manager level. Our measure of network centrality for manager j is again normalized relative to the cross-sectional mean in the same period, i.e., $NET_{jt} = N_{jt}/\overline{N}_t$, where N_{jt} is the degree centrality measure for manager j at time t, and \overline{N}_t is the cross-sectional average (across i and j) at time t. Centrality is always measured ex-ante, i.e., prior to the return measurement period. To capture possible scale effects of network centrality, we also consider an interaction term between network centrality and manager size, $NET_{jt}Msize_{jt}$. Notice that it is the variation *over time* in network centrality that identifies its effect on risk-adjusted performance.

¹¹We include a North American market return factor separately due to the evidence in Timmermann and Blake (2005) that UK pension funds considerably overweighted this market in their international equity portfolio.

For each asset class, Table 3 presents results from panel regressions of the form¹²

$$r_{ijt}^{adj} = a_{ij} + b_t + \lambda_1 size_{ijt} + \lambda_2 M size_{jt} + \lambda_3 NET_{jt} + \lambda_4 NET_{jt} M size_{jt} + \varepsilon_{ijt}.$$
(6)

This regression allows us to study the effect of network centrality, NET_{ijt} , (measured within each asset class) while controlling for variations in fund-manager or manager size and allowing for fund-manager and time fixed effects. The fund-manager effect controls for abnormal performance due to skills in matching manager j with fund i. For each asset class we show results with the network-size interaction term switched off and on (odd and even columns, respectively), along with regressions based on the total, manager-only and consultant-only network measures.

First consider the effect of size on risk-adjusted performance shown in rows 1 and 2 in Table 3. This varies quite a bit across the three asset classes. For UK equities (columns 1-6), manager size is significantly negatively correlated with risk-adjusted returns, while fund-manager size does not appear to matter. For UK bonds (columns 7-12), fund-manager as well as manager size are both negatively correlated with performance, though only the former is borderline significant. Finally, for international equities (columns 13-18) we find that fund-manager size is strongly negatively correlated with performance while the manager size does not matter.

Next, consider the effect of network centrality on risk-adjusted performance in UK equities (columns 1-6). The first column, which is based on both manager- and consultant centrality, shows that there is a positive and significant effect of centrality on risk-adjusted performance. The coefficient is quite large, suggesting that a one-unit increase in centrality raises expected risk-adjusted returns by 0.35% per annum. Note that this result holds after controlling for both fund size and manager size. In column two we find a strongly positive effect of including a centrality-manager size interaction term. The significance of the interaction term suggests that the performance of large managers is more sensitive to network centrality than that of small managers and that a central position in the network helps managers cushion the otherwise strongly negative effect of size on performance. The interaction term weakens the direct effect of the network term but strengthens the results in the sense that the joint significance of the two network terms is now significant with a *p*-value less than one percent.

 $^{^{12}}$ For simplicity we suppress reference to the asset class in the notation.

Columns three and four show results for regressions that use the manager-only network centrality measure. We find borderline evidence that manager centrality is related to risk-adjusted performance (column three) with a p-value of 6%. However, the centrality-manager size interaction term is once again highly significant. The results are even stronger for the consultant-only centrality measure which generates a positive and significant (p-value of 0.03) coefficient that remains significant once the centrality-size interaction term is included.

These results do not reveal whether network connections established through managers, through consultants, or through both, matter most for risk-adjusted investment performance. To address this point, we next undertake a two-step encompassing regression. Specifically, we obtain the residuals from (6) based on the manager-only centrality measure. By construction, these residuals are orthogonal to the manager-only centrality terms. We then regress these residuals on the consultant-only network terms and perform a joint significance test. Rejection of this test (i.e., a low p-value) suggests that the consultant-only centrality measure helps explain part of the risk-adjusted return performance that the manager-based centrality measure does not explain. For UK equities, we find a p-value of 0.09, suggesting significant evidence (at the 10% level) that centrality obtained through the consultant networks helps explain part of the return performance that is not explained by the manager centrality measure.

Analogously, we obtain the residuals from (6) based on the consultant-only centrality measure, project these residuals on the manager-only network terms and perform a joint significance test. For this case, a low *p*-value suggests that the manager-based centrality measure helps explain part of the risk-adjusted return performance that the consultant-only centrality measure does not capture. In this case, with a *p*-value of 0.24 we find no evidence that manager-based centrality helps explain excess return performance not explained by consultant centrality.

Turning to UK bonds, results for which are reported in columns 7-12 in Table 3, the evidence strongly suggests a positive relation between network centrality and risk-adjusted return performance, regardless of which of the three centrality measures (total, manager-only, or consultant-only) is used. Interestingly, the size of the centrality effect is strongest for the consultant-only network. In contrast to the results for UK equity, for UK bonds the coefficient on the centrality-size interaction term is negative and significant in two out of three cases. For all three centrality measures the (marginal) effect of centrality continues to be positive and significant even after including the size-centrality interaction term. This suggests that centrality is always associated with better investment performance for UK bond managers, although the ability to take advantage of a central network position is reduced for the largest UK bond managers. With a p-value of 0.00, the encompassing test strongly suggests that the consultant centrality measure helps explain return performance in UK bonds not explained by manager centrality. Conversely, there is no evidence to suggest the converse, i.e., that manager centrality explains excess returns over and above that identified through the consultant centrality measure—the associated p-value is 0.80.

In the case of international equities (columns 13-18), we find little evidence that network centrality matters to investment performance, except perhaps for some mild evidence that a larger consultant centrality-size interaction term is associated with better performance (last column). In this case, both of the encompassing tests come out insignificantly suggesting that neither of the consultant or manager-based network measures encompasses the other.

Overall, these results suggest a positive relation between a fund's network centrality and its ability to generate risk-adjusted performance in the UK equity and bond markets, i.e., more centrally positioned managers tend to be those with the best investment performance. Conversely, the centrality of our UK network of managers does not appear to matter for the funds' investment performance in international equities.

A plausible explanation for these findings is that while better UK network connections can be exploited to generate better performance in domestic asset markets, they do not easily translate into information that can be used to achieve better investment performance in foreign markets.

3.3 Role of investment consultants

The vast financial literature on investment performance focuses either on the performance of individual funds–mostly mutual funds, but also hedge funds and pension funds–or on fund managers.¹³

To our knowledge, no prior study has considered the role investment consultants play in the

¹³Chevalier and Ellison (1999), Baks (2001) and Ding and Wermers (2012) are examples of studies focusing on the relation between portfolio manager characteristics and investment performance.

investment process, let alone how they influence outcomes. This is an important omission since consultants play an important role in pension funds' choice of investment manager. A key issue is, therefore, whether they actually affect the funds' investment performance.

To address this question, we focus on the four largest consultants for which we have a sufficient number of observations to be able to compute, and compare, investment performance. Specifically, for each fund-consultant pairing we compute the mean return in UK equities in excess of the associated benchmark and divide this by the standard deviation of the associated residual so as to get an estimate of the information ratio (IR).¹⁴ Thus, if a fund is initially advised by consultant 2 followed by a switch to consultant 11, the fund's performance record during the first period will be allocated to consultant 2, while the record during the latter period is assigned to consultant 11.

Figure 9 shows kernel density plots of the distribution of IR estimates for the four largest consultants. The figure indicates notable differences in the distribution of information ratios across consultants. For example, consultant 11 has a distribution that is further to the left-indicating worse performance-than that of the other consultants. Consultant one and two also appear to have higher "upside" performance, marked by slightly heavier right tails. Consultants one and two have average information ratios of 0.11, consultant three has an IR of 0.051 while consultant 11 has an IR of -0.07.

To test if these impressions translate into statistically significant differences, we next undertake a non-parametric permutation test which provides pairwise comparisons of the population of funds advised by different consultants.¹⁵ Table 4 presents the results from this test. Small *p*-values indicate significant evidence that one manager is better than the other. Consultants 1, 2, and 3 all produce higher mean IRs than consultant 11, while consultants 1 and 2 also better the performance of consultant 3 (at the 10% significance level). There is little to distinguish the performance of consultants 1 and 2. These results suggest that consultants influence the performance of the funds they advise.

¹⁴This is a more reliable measure than the Sharpe ratio-mean excess return relative to the T-bill rate, divided by the volatility of excess returns-which is sensitive to the period over which the fund-consultants are in existence. For example, consultant 11 has a high average Sharpe ratio for UK equities simply because it is dominated by observations prior to 2000-a period during which UK equity returns were unusually high.

¹⁵The test pools the populations of funds across two consultants and then reassigns them, at random and without replacement, to the two consultants. It then computes the difference between the mean information ratios for the two consultants and compares this to the difference observed for the funds actually advised by the consultants. The p-value of the test is the proportion of permutations with a difference at least as large as the actual difference.

4 Fund flows and network centrality

We next ask whether managers' centrality in the network affects flows of money into the funds that they manage. In this analysis we aggregate assets under management across all asset classes so as to avoid that the results are contaminated by any shifts in asset allocation undertaken by the manager for reasons unrelated to network centrality. Moreover, we consider the results at the manager, rather than the fund-manager level. Because we are analyzing defined-benefit pension fund accounts, the flows into a given fund-manager account is likely to be determined by factors extraneous to the pairing's centrality in the network such as numbers of employees and their contributions. Conversely, managers may get new client accounts if they are perceived as being capable of delivering superior investment performance.

Specifically, in this analysis we first generate a fund-flow variable for manager i over the course of quarter t as follows:

$$Flow_{jt+1} = \left(\frac{SMV_{j,t+1} - SMV_{j,t}}{SMV_{j,t}} - R_{jt+1}\right)SMV_{j,t},\tag{7}$$

where $SMV_{j,t}$ is the starting market value of manager j's total asset holdings at the end of quarter t and R_{jt+1} is the return generated by the manager during quarter t + 1.

We regress the manager flow variable defined in (7) on the lagged flow, manager size, $Msize_{jt}$, past returns over the previous year, \bar{R}_{jt} , network centrality, NET_{jt} , along with centrality-size and centrality-return interaction terms, allowing for time and manager fixed effects:

$$Flow_{jt+1} = a_j + c_t + \beta_1 Flow_{jt} + \beta_2 M size_{jt} + \beta_3 \bar{R}_{jt} + \beta_4 N E T_{jt} + \beta_5 M size_{jt} N E T_{jt} + \beta_5 \bar{R}_{jt} N E T_{jt} + \varepsilon_{jt}.$$
(8)

Results from this regression are shown in panels A-C in Table 5. For UK equities (Panel A), past flows appear to be uncorrelated with future flows, while manager size is strongly negatively related to future flows. Past return performance does not appear to matter on its own. However, there is clear evidence that more central managers attract larger inflows of funds, regardless of whether we use the total, manager-only or consultant-only centrality measures. The effect of centrality on fund flows seems unrelated to manager size (columns 6 and 8) but is boosted by consultant size as shown by the positive and significant interaction terms (columns 10 and 12.) Moreover, we also find positive and significant coefficients on the centrality-past return interaction terms, suggesting that higher past returns are exploited by the most central managers to generate larger inflows.

For UK bonds (Panel B), again we find positive and borderline significant coefficients on the centrality measures in the flow regressions that use the manager-only or total centrality measures, though not for the consultant-based centrality measure. Manager size continues to be negatively associated with fund flows and the centrality interaction terms are mostly insignificant, the one exception being the consultant-only centrality-size interaction coefficient which is small but positive.

Results for International equity are similar to those obtained for UK equity fund flows: fund flows are negatively associated with manager size, but positively related to manager and consultant centrality. Moreover, the centrality-past returns and centrality-size interaction terms are positive and predominantly significant, suggesting that network centrality provides the greatest benefits to the largest funds with the highest past returns.

5 Risk-taking and Network Centrality

Section 3 established a positive association between fund-manager centrality and risk-adjusted investment performance for UK equities and UK bonds, but not for international equities. We next address whether centrality affects managers' willingness to take risk and the consequences of such actions.

Network centrality could affect managers' risk-taking for at least two separate reasons. First, if more centrally placed managers have access to more precise information, they may be willing to take what appears to outsiders to be riskier bets. Second, if more centrally placed managers are less likely to be fired for a given level of investment performance (as we find in a later section), then they should also be willing to take riskier bets. There is likely to be a size effect on managers' risk taking as well, however, as larger managers will find it more difficult to deviate from the market benchmark due to a greater market impact of their trades and less maneuverability compared with that of smaller managers.

5.1 Idiosyncratic Risk and Network Centrality

We perform our analysis by proxying for the unobserved level of risk by means of the level of idiosyncratic risk taken by a fund manager. Specifically, using equations (3)-(5), we first extract an estimate of the fund-manager pairing's idiosyncratic risk, $|\hat{\varepsilon}_{ijt}|$. Note that if these residuals are drawn from a Gaussian distribution, then $E[|\hat{\varepsilon}_{ijt}|] = \sqrt{2/\pi} st dev(\hat{\varepsilon}_{ijt})$, thus justifying this particular proxy for risk. It should be recognized, however, that this is clearly a noisy measure of risk as it is based on a single observation for every period.

To estimate how funds' idiosyncratic risk taking is affected by network centrality, we next estimate panel regressions similar to (6):

$$|\hat{\varepsilon}_{ijt}| = a_i + b_j + c_t + \lambda_1 size_{ijt} + \lambda_2 M size_{jt} + \lambda_3 NET_{ijt} + \lambda_4 NET_{ijt} M size_{jt} + \varepsilon_{ijt}.$$
 (9)

Results from this regression are presented in Table 6 which uses the same layout as Table 3. First, notice that fund-manager size as well as manager size have a strongly negative effect on risk-taking across all three asset classes and across different specifications: Larger funds and larger managers take on less active risk and mirror the benchmarks more closely for each of the three asset classes.

For UK equities (columns 1-6) there is modest evidence that funds with higher consultant-only centrality take on more risk than less central funds. Conversely, the corresponding coefficients for the total and manager-only centrality measures are negative and insignificant. Across the three centrality measures, the centrality-size interaction term generates a negative and highly significant coefficient, however. This suggests that large and centrally placed managers are even less willing to take on idiosyncratic risk than what is explained by size alone.

Similar results are found for UK bond managers–again the centrality-size interaction term is negative and strongly significant while the centrality measure on its own is only significant for one of three cases, this time generating a negative coefficient for the consultant-only network centrality measure.

For International equities we find very different results. For this asset class, the coefficient on network centrality is now positive and statistically significant in two of three cases and the results are now weakest for the consultant-based centrality network. Similarly, the centrality-size interaction term now generates a positive coefficient that remains significant across all three centrality measures. Whereas large, centrally connected UK equity and UK bond managers tend to take on less risk, we thus find that centrally positioned international managers are willing to take riskier bets.

A possible explanation of this evidence is the finding in previous studies such as Kang and Stulz (1997) that foreign investors act as if they are at an informational disadvantage when investing abroad. Our finding is consistent with more central managers—who are likely to have international offices and networks of their own—being less cautious when investing abroad than the managers who are less well connected.

5.2 Network Centrality and Hazards of Firing

Network centrality does not only affect the information flowing to and from a particular manager or consultant; it can also affect the manager or consultant's incentives. This can happen through its effect on flows of funds into and out of the funds under the manager's (consultant's) control and thus the manager's remuneration which is likely to depend on the asset base; we analyzed this effect in Section 4. It can also happen through its effect on the probability that the manager or consultant is fired by a client. To address this second channel, we next analyze whether the probability that a manager or consultant is fired is affected by his network centrality.¹⁶

5.2.1 Modeling the Hazard of Being Fired

To see whether a fund manager's or consultant's probability of being fired is influenced by his position (centrality) in the network, we estimate hazard rate models. The hazard rate (h) measures the probability of being fired next period, conditional on having survived up to the present time. To avoid having to impose restrictions on how the baseline hazard rate depends on the duration (d) of the relation between a manager and the pension fund, we use the Cox semi-parametric regression approach. This allows the effect of the 'age' of the fund-manager relationship-denoted the baseline hazard rate, $h_0(d)$ -to be estimated nonparametrically, while the effect of fund-manager variables, x_{ijt} , on the hazard rate is modeled parametrically.

¹⁶Previous studies have analyzed the factors influencing the likelihood of termination for mutual fund managers. Chevalier and Ellison (1999) report that young managers face a higher risk of being fired following poor risk-adjusted performance. Khorana (1996) finds that underperforming managers with decreasing inflows also face a higher probability of being fired.

Specifically, letting $h(d_{ijt})$ be the hazard rate for fund-manager i, j at time t and $h_0(d_{ijt})$ be the baseline hazard rate as a function of the duration of the fund-manager's tenure at time t, we estimate the following semiparametric Cox regression model

$$h(d_{ijt}) = h_0(d_{ijt}) \exp(\beta' x_{ijt}).$$
⁽¹⁰⁾

Our model allows the manager's hazard rate to depend on four factors, x_{it} . First, we consider the effect of the duration of manager j's relation with pension fund i at time t, d_{ijt} , measured in quarters. This maps into the baseline hazard, $h_0(d_{ijt})$. We present results for this variable graphically. An upward-slopping curve indicates that the manager's risk of getting fired increases as the length of his contract with a pension fund gets extended, while a downward-slopping curve suggests that the manager is less likely to get fired, the longer he has been with a particular pension fund.

Second, past risk-adjusted returns could affect the probability that the manager is fired and so we include a return measure, \bar{R}_{it}^{adj} , which measures the manager's return over the previous four quarters. Returns are risk-adjusted, i.e., adjusted for a fund-manager's exposure to the Fama-French (1993) size and value common risk factors, augmented by the Carhart (1997) momentum factor (SMB_t, HML_t , and MOM_t , respectively). The hypothesis here is that higher past risk-adjusted returns should reduce the chance of getting fired.

Third, we control for manager size, $MSIZE_{jt}$, measured at the beginning of each quarter. We found earlier that this matters for both return performance and fund flows and so it is natural to expect this variable also to be important for managers' prospects of getting fired. Here the hypothesis is that, after controlling for past return performance, large, established managers are less likely to get fired than smaller managers.

Our final covariate is the centrality measure, NET_{jt} . To see if it matters whether network centrality is established through consultants or directly between managers, we also consider these centrality measures separately. The hypothesis is that the more central a manager is within the network, the less likely he is to get fired. In summary, our regression model for the hazard rate takes the following form:

$$h(d_{ijt}, \bar{R}_{ijt}, Msize_{ijt}, NET_{ijt}) = h_0(d_{ijt}) \exp(\beta_1 \bar{R}_{ijt} + \beta_2 Msize_{ijt} + \beta_3 NET_{ijt}) + \varepsilon_{ijt}.$$
 (11)

5.2.2 Results for Managers

Figure 10 (top panels) plots the (smoothed) baseline hazard rates for the three asset classes. These show how the probability that a manager is fired in the subsequent quarter varies with the duration of the fund-manager contract. For all three asset classes, the hazard rate increases systematically in the duration, quadrupling from around 0.2 - 0.3% per quarter for managers with a tenure of 10 quarters to 0.8% - 1.2% per quarter for managers with a tenure of 70 quarters.

Panel A in Table 7 reports estimation results for the model in (11). The hazard rate, i.e., the risk that the manager is fired by a client, is significantly negatively related to past performance for both UK equities and international equities, but generates an insignificant coefficient for UK bonds. Higher past (risk-adjusted) performance is thus associated with a reduced probability that a manager will be fired by the fund.

We estimate a large negative, and highly significant, coefficient on manager size, suggesting that large managers face a lower probability of being fired. Moreover, the estimated coefficient is largely the same across the three asset classes and is robust to the included measure of centrality.

Turning to network centrality, all specifications generate negative coefficients on this measure. The coefficients are significant at the 5% apart from one case (UK bonds, manager-only centrality) that is significant at the 10% level. Thus, more central managers appear to face a greatly reduced chance of being fired compared with more peripheral managers. The effect is strongest for UK and International equities.

These results establish a strong case that managers' network centrality negatively affects their probability of being fired. Besides the advantages that centrality offers managers in terms of higher inflows and better past performance, this suggests that more central managers also are "safer", i.e., they stand a lower chance of being fired.

5.2.3 Results for Consultants

We next undertake a similar hazard rate analysis for the consultants as that undertaken for the managers. Note, however, that once we use consultants' firing as the dependent variable, we can only construct the overall degree measure and so omit two of the three centrality measures.

Figure 10 (bottom panels) plots the baseline hazard rate for our sample of consultants. The figure again shows broad evidence of a hazard rate that increases in the duration of the fund-consultant relation. Note, however, that the baseline level is somewhat lower than for the managers and ranges from around 0.2% per quarter for newly formed relations to 0.8% per quarter for relations longer than 15 years (60 quarters).

Panel B of Table 7 shows parameter estimates from applying (11) to the consultant data. Past average return performance no longer appears to be a significant predictor of firing events for UK equities and UK bonds, although it generates a negative and significant coefficient for International equities. Consultant size remains strongly negatively related to the firing probability in all three asset classes. Interestingly, consultant centrality is, once again, a negative and highly significantly predictor of future firings in all three asset classes. Thus we find that, for managers and consultants alike, network centrality helps reduce the risks of getting fired.

6 Size and Network Centrality: Granger Causality Tests

Table 2 shows that a manager's size and network centrality are positively connected. Large managers are likely to manage the assets of more clients and so this finding does not come as a surprise. While it can be difficult to formally test if size causes centrality or vice versa, more limited tests of whether one variable precedes the other one are feasible through Granger causality tests.

To implement such Granger causality tests, we first obtain the centrality measure for each manager at each point in time. Similarly, we compute the size of each manager by aggregating investments in all the funds (and asset classes) managed by the manager. We then regress changes in log-size on its own lag and the lag of changes in degree centrality. Because the lagged size and lagged centrality measures are not exogenous, we instrument these variables by using their own lags using instrumental variable estimation. Specifically, we follow the procedure for Granger causality tests in a panel setting developed by Holtz-Eakin, Newey, and Rosen (1988). To this end, consider the simple panel model:

$$y_{it} = \lambda_0 + \sum_{l=1}^m \lambda_l y_{it-l} + \sum_{l=1}^m \delta_l x_{it-l} + u_{it}.$$
 (12)

The model in (12) is a simple pooled OLS that imposes the constraint that the underlying structure is the same for each cross-sectional unit. This assumption can be relaxed either by introducing an individual specific intercept—so as to allow for individual heterogeneity in the levels of x and y—or by allowing the variance of the innovation in (12) to vary with the cross-sectional unit so as to capture individual heterogeneity in the variability of x and y.

Working with a panel of data rather than individual time-series offers key advantages. We can allow the coefficients on the lags to vary over time and the large number of cross-sectional units does not require the vector autoregression to satisfy the usual conditions that rule out unit roots or even explosive roots. Exploiting these advantages, Holtz-Eakin et al. (1988) propose a model that allows for individual effects and non-stationarities:

$$y_{it} = \lambda_{0t} + \sum_{l=1}^{m} \lambda_{lt} y_{it-l} + \sum_{l=1}^{m} \delta_{lt} x_{it-l} + \Psi_t f_i + u_{it},$$
(13)

where f_i is an unobserved individual effect and the coefficients $\lambda_{0t}, \lambda_{1t}, ..., \lambda_{mt}, \delta_{1t}, ..., \delta_{mt}, \Psi_t$ are the coefficients of the linear projection of y_{it} on a constant, past values of y_{it} and x_{it} and the individual effect f_i . Implicit in equation (13) is that for each period, t, the projection of y_{it} on the entire past depends only on the past m observations.

Our analysis of the relation between fund size and network centrality uses the specification

$$y_{it} = \lambda_0 + \sum_{l=1}^m \lambda_l y_{it-l} + \sum_{l=1}^m \delta_l x_{it-l} + \Psi f_i + u_{it}.$$

First-differencing this model yields

$$y_{it} - y_{it-1} = \sum_{l=1}^{m} \lambda_l (y_{it-l} - y_{it-l-1}) + \sum_{l=1}^{m} \delta_l (x_{it-l} - x_{it-l-1}) + v_{it},$$
(14)

where $v_{it} = u_{it} - u_{it-1}$.

Estimation

To estimate the model, define $N \times 1$ vectors of observations on the various units at a given time period, $\mathbf{Y}_t = (Y_{1t}, ..., Y_{Nt})'$ and $\mathbf{X}_t = (X_{1t}, ..., X_{Nt})'$. Let $\mathbf{W}_t = (\mathbf{e}_N, \mathbf{Y}_{t-1}, ..., \mathbf{Y}_{t-m}, \mathbf{X}_{t-1}, ..., \mathbf{X}_{t-m})$ be the matrix of regressors, where \mathbf{e}_N is an $N \times 1$ vector of ones. Further, let $\mathbf{V}_t = (v_{1t}, ..., v_{mt})'$ be the vector of transformed disturbance terms and let $\mathbf{B} = (a, \lambda_1, ..., \lambda_m, \delta_1, ..., \delta_m)'$ be the vector of coefficients. Then we can write (14) as:

$$\mathbf{Y}_t = \mathbf{W}_t \mathbf{B} + \mathbf{V}_t. \tag{15}$$

Stacking the observations for each time period, we can simplify this to a system

$$\mathbf{Y} = \mathbf{W}\mathbf{B} + \mathbf{V}.\tag{16}$$

Finally, defining a set of instrumental variables, \mathbf{Z} , we estimate \mathbf{B} from the equation

$$\mathbf{Z}'\mathbf{Y} = \mathbf{Z}'\mathbf{W}\mathbf{B} + \mathbf{Z}'\mathbf{V}.$$
 (17)

This specification makes it easy to test whether the coefficients on the x-variables are jointly equal to zero by imposing simple linear restrictions and then computing the likelihood ratio test comparing the restricted and unrestricted model.

Our implementation uses one-step GMM estimation and the Arellano-Bond estimator and limits the instruments to a maximum of 16 lags.¹⁷ We separately consider the total, manager- and consultantbased centrality measures.

6.1 Empirical Findings

Table 8 presents the outcome of the Granger causality tests described above as applied to our data. Panel A uses the centrality measure as the dependent variable, while lagged size and lagged centrality

¹⁷This is due to the fact that we have 81 observations in the time-series and the number of instruments would become unmanageably large otherwise.

is used as independent variables. In all but one instance the lagged size fails to significantly predict centrality, leading to the conclusion that size does not Granger cause network centrality. As expected, lagged centrality predicts current centrality, consistent with the persistence in the centrality measures revealed in plots such as Figure 8.

Panel B of Table 8 performs the reverse regression, regressing current size on lagged centrality and past size. Here we find that centrality strongly (and positively) predicts future size, after controlling for past size. Thus, network centrality Granger causes size, but not the reverse. This result is consistent across our three different measures of network centrality and across asset classes and so are very strong. Moreover, the results are robust to the number of lags chosen.

The conclusion from these results is that network centrality adds a novel dimension to our understanding of managers' investment performance, risk-taking behavior, and fund flows. Moreover, network centrality, though positively related to size, is clearly not subsumed by size. In fact, although size and network centrality are positively correlated, size generally has a negative effect on investment performance and fund inflows while conversely network centrality is associated with better risk-adjusted performance and higher inflows.

7 Conclusion

Financial systems are intricate, highly interconnected networks in which the relations between institutional clients, fund managers, and investment advisors (consultants) evolve dynamically in a way that reflects past performance which, in turn, will affect future performance. How information flows between clients and investment managers is important for both investment performance, flows of funds, and the incentives and risk-taking behavior of fund managers. No prior study has been able to address this question empirically.

This paper uses a unique data set to shed light on these questions. First, we document how funds and fund-managers are connected and how such connections evolve over time. We distinguish between network connections established through managers versus connections established through consultants. Next, we show that a fund-manager's (relative) degree of centrality in a network positively affects riskadjusted returns and growth in assets under management and that this effect is particularly strong for large fund managers, even after controlling for size. In fact, a central position in the network seems to be important for reducing the large negative effect a manager's size ordinarily has on his return performance and future growth.

These results suggest that there is path dependence in fund managers' performance: the ability of a manager to establish a central position in the network of institutional investors matters to his future success as measured by risk-adjusted returns and growth in assets under management. In turn these measures of performance affect the managers' ability to exploit their central position in the network. Once a central position has been established and the manager has grown large, the manager tends to reduce risk-taking behavior and reduces the chances of getting fired by institutional clients.

References

- Avramov, D. and R. Wermers, 2006, Investing in Mutual Funds when Returns are Predictable. Journal of Financial Economics 81, 339-377.
- [2] Baks, Klaas, 2001, On the Performance of Mutual Fund Managers. Working Paper, Emory University
- Banegas, A., B. Gillen, A. Timmermann, and R. Wermers, 2013, The Cross-section of Conditional Mutual Fund Performance in European Stock Markets. Journal of Financial Economics 108, 699-726.
- [4] van Binsbergen, Jules H., Michael W. Brandt, and Ralph S. J. Koijen, 2008, Optimal decentralized investment management, Journal of Finance 63, 1849–1894.
- [5] Blake, D., A.G. Rossi, A. Timmermann, I. Tonks, and R. Wermers, 2013, Decentralized Investment Management: Evidence from the Pension Fund Industry. Journal of Finance 68, 1133-1178.
- [6] Bollen, N.P.B., and J. A. Busse, 2004, Short-Term Persistence in Mutual Fund Performance. Review of Financial Studies 18, 569-597.
- [7] Carhart, M., 1997, On Persistence in Mutual Fund Performance. Journal of Finance 52, 57-82.
- [8] Chevalier, Judith, and Glenn Ellison, 1999a, Are Some Mutual Fund Managers Better than Others? Cross-Sectional Patterns in Behavior and Performance, Journal of Finance, 54(3), 875-899.
- [9] Chevalier, Judith, and Glenn Ellison, 1999b, Career Concerns of Mutual Fund Managers. Quarterly Journal of Economics, 114, 389-432.
- [10] Cohen, L., A. Frazzini, C. Malloy, 2008, The small world of investing: board connections and mutual fund returns. Journal of Political Economy 116, 951–979.
- [11] Cremers, K.J.M., and A. Petajisto, 2009, How active is your fund manager? A new measure that predicts performance. Review of Financial Studies 22, 3329–3365.

- [12] Ding, B., and R. Wermers, 2012, Mutual fund performance and governance structure: The role of portfolio managers and boards of directors. Manuscript, SUNY and Albany and University of Maryland.
- [13] Holtz-Eakin, D., W. Newey, and H.S. Rosen, 1988, Estimating vector autoregressions with panel data. Econometrica 56, 1371-1396.
- [14] Kacperczyk, M., S. van Nieuwerburgh, and L. Veldkamp, 2013, Time-varying Mutual Fund Manager Skill. Journal of Finance (forthcoming).
- [15] Kang, J. K., and R. Stulz, 1997, Why is There a Home Bias? An Analysis of Foreign Portfolio Equity Ownership in Japan. Journal of Financial Economics 46, 3–28.
- [16] Khorana, Ajay, 1996, Top Management Turnover: An Empirical Investigation of Mutual Fund Managers. Journal of Financial Economics 40, 403-427.
- [17] Khorana, Ajay, 2001, Performance Changes Following Top Management Turnover: Evidence from Open-End Mutual Funds. Journal of Financial and Quantitative Analysis 36, 371-393.
- [18] Kosowski, R., Timmermann, A., Wermers, R., & White, H., 2006, Can mutual fund "Stars" really pick stocks? New evidence from a bootstrap analysis. Journal of Finance, 61(6), 2551-2595.
- [19] Mamaysky, H., Spiegel, M., Zhang, H., 2008. Estimation the dynamics of mutual fund alphas and betas. Review of Financial Studies 21.
- [20] Sharpe, William F., 1981, Decentralized investment management, Journal of Finance 36, 217– 234.
- [21] Timmermann, Allan, and David Blake, 2005, International asset allocation with time-varying investment opportunities, Journal of Business 78, 71–98.



Figure 1. Evolution of the Number of Mandates by Consultants.

This figure plots the number of mandates by consultants over the period 1984 to 2004. The results for UK equities are reported in Panel A, the results for UK bonds are reported in Panel B and the results for international equities are reported in Panel C.



Figure 2. Relative Size of Consultants over Time.

This figure plots the relative size of consultants over the period 1984 to 2004. The total size of each asset class is normalized to one in every period and for each consultant we report the proportion of assets managed relative to the total size of the asset class. The analysis is conducted for UK equities in Panel A, for UK bonds in Panel B and for international equities in Panel C.





(C) year : 2004

This figure plots the network connections in UK equities at three points in time during our sample, namely 1984, 1994, and 2004. The red circles represent individual managers, while the black diamonds in the horizontal row represent the 12 consultants. Next to each node is shown the code of the manager or consultant. Managers whose nodes are shown above the consultants are only connected through the consultants, while the managers whose nodes fall below the consultants are connected with at least one other manager.



Figure 4. Network connections in UK bonds asset class in the UK pension fund industry

(C) year : 2004

This figure plots the network connections in UK bonds at three points in time during our sample, namely 1984, 1994, and 2004. The red circles represent individual managers, while the black diamonds in the horizontal row represent the 12 consultants. Next to each node is shown the code of the manager or consultant. Managers whose nodes are shown above the consultants are only connected through the consultants, while the managers whose nodes fall below the consultants are connected with at least one other manager.

Figure 5. Network connections in international equities asset class in the UK pension fund industry



(C) year : 2004

This figure plots the network connections in international equities at three points in time during our sample, namely 1984, 1994, and 2004. The red circles represent individual managers, while the black diamonds in the horizontal row represent the 12 consultants. Next to each node is shown the code of the manager or consultant. Managers whose nodes are shown above the consultants are only connected through the consultants, while the managers whose nodes fall below the consultants are connected with at least one other manager.





This figure plots the network connections between consultants and managers in 2004 for UK equities (Panel A), UK bonds (Panel B) and international equities (Panel C). Each green sphere represents a consultant. The size of the spheres is a function of the relative degree centrality of the manager or the consultants. The lines connecting the various spheres represent manager-to-manager connections and manager-to-consultant connections.

This figure plots a simple example of a network with 7 nodes and reports the degree centrality values for each node in the network. Degree centrality values are computed using equation (1) in the main body of the paper.

Figure 8. Average network centrality in the UK pension fund industry

This figure plots the time series of the average degree centrality, the average degree centrality computed using the managers' network only and the average degree centrality computed using the consultants' network only. The centrality measures are computed using network connections in UK equities only in Panel (A), network connections in UK bonds only in Panel (B) and network connections in international equities only in Panel (C) and network connections across all asset classes in Panel (D). In each panel, each average centrality measure " CM_t " is standardized as follows

$$S_{-}CM_t = \frac{CM_t - MEAN(CM_t)}{STDEV(CM_t)},$$

where $MEAN(CM_t)$ is the time-series mean of the average centrality measure CM_t and $STDEV(CM_t)$ is its standard deviation.

Figure 9. Information Ratios of Consultants Across Funds Managed

This figure plots the kernel density estimates of the information ratios in UK equities associated with the four consultants with the largest number of mandates, i.e. Consultant 1, Consultant 2, Consultant 3 and Consultant 11. The information ratio of consultant i in fund j is computed as

$$IR_{i,j} = \frac{\frac{1}{(t_{max} - t_{min} + 1)} \sum_{t_{min}}^{t_{max}} (r_{i,j,t} - r_{m,t})}{\sqrt{\frac{1}{(t_{max} - t_{min})} \sum_{t_{min}}^{t_{max}} \left((r_{i,j,t} - r_{m,t}) - \overline{(r_{i,j,\cdot} - r_{m,\cdot})} \right)^2}}$$

where t_{min} (t_{max}) is the first (last) observation of each consultant-fund pairing, $r_{i,j,t}$ is the excess return of consultant *i* in fund *j* at time *t* and $r_{m,t}$ is the excess return on the UK market portfolio at time *t*. We drop from the sample the consultant-manager pairings that have less than 12 observations. The information ratios are annualized by multiplying $IR_{i,j}$ by 2. Figure 10. Baseline Hazard Rates for Managers' and Consultants' Survival Analysis

Managers' Baseline Hazard Rates

UK equities, panels B and E display the results for UK bonds and panels C and F display the results for international equities. All hazard functions are estimated This figure plots baseline hazard functions for fund-managers in Panels A through C and consultants in Panels D through F. Panel A and D display the results for non-parametrically.

(E) UK Bonds

(D) UK Equities

(F) International Equities

			Panel A	A. UK Equ	ities			
Year 1984 1994 2004	N. of Fund-Managers 1204 1420 1053	N. of Funds 955 1044 630	N. of Managers 113 112 82	$Net = 1 \\ 35 \\ 34 \\ 22$	$2 \leq Net \leq 5$ 35 42 23	$\begin{split} 6 \leq Net \leq 10 \\ 15 \\ 14 \\ 17 \end{split}$	$\begin{array}{c} 11 \leq Net \leq 20\\ 20\\ 12\\ 10\end{array}$	$Net \geq 21$ 8 10 10
			Panel	B. UK Boı	shr			
Year 1984 1994	N. of Fund-Managers 1165 745	N. of Funds 943 652	N. of Managers 109 96	Net = 1 37 36	$2 \leq Net \leq 5$ 32 41	$6 \leq Net \leq 10$ 15 8	$\begin{array}{c} 11 \leq Net \leq 20\\ 20\\ 10\end{array}$	$Net \geq 21$ 5 1
2004	817	612	61	19	22	9	10	4
			Panel C. Int	ternational	Equities			
Year 1984 1994	N. of Fund-Managers 1135 1354	N. of Funds 911 1019	N. of Managers 108 118	Net = 1 34 37	$2 \leq Net \leq 5 \ 35 \ 35$	$6 \leq Net \leq 10$ 11 17	$\begin{array}{c} 11 \leq Net \leq 20\\ 21\\ 21\\ 31\end{array}$	$Net \geq 21$ 7 7
2004	956	627	89	22	32	14	12	- 0

Table 1. Number of Funds. Managers and their Connections Across Time and Asset Classes

This table presents summary statistics for the number of fund-manager pairings, number of funds, and number of managers in 1984, 1994 and 2004. For the same years, it also reports the number of managers with number of managers with network connections between 2 and 5 ($2 \le Net \le 5$), the number of managers with number with network connections between 11 and 20 ($11 \le Net \le 21$), and the number of managers with number with network connections between 11 and 20 ($11 \le Net \le 21$), and the number of managers with number with network connections between 11 and 20 ($11 \le Net \le 21$), and the number of managers with number with network connections between 11 and 20 ($11 \le Net \le 21$), and the number of managers with number with network connections greater than 20 ($Net \ge 21$). The analysis is conducted for UK equities in Panel A, UK bonds in Panel B and international equities in Panel C.

Table 2. Correlation between centrality and size measures

Panel A. Results Across Asset Classes

	NET	NETM	NET_C	SIZE	M_SIZE
NET	1.000				
NET_M	0.998	1.000			
NET_C	0.840	0.802	1.000		
SIZE	-0.008	-0.008	-0.014	1.000	
M_SIZE	0.653	0.647	0.552	0.092	1.000

Panel B. Results for UK Equities

	NET	NET_M	NET_C	SIZE	M_SIZE
NET	1.000				
NET_M	0.996	1.000			
NET_C	0.866	0.817	1.000		
SIZE	-0.017	-0.018	-0.012	1.000	
M_SIZE	0.637	0.634	0.564	0.092	1.000

Panel C. Results for UK Bonds

	NET	NET_M	NET_C	SIZE	M_SIZE
NET	1.000				
NETM	0.985	1.000			
NET_C	0.872	0.787	1.000		
SIZE	-0.018	-0.021	-0.005	1.000	
M_SIZE	0.542	0.533	0.478	0.092	1.000

Panel D. Results for International Equities

	NET	$NET_{-}M$	NET_C	SIZE	M_SIZE
NET	1.000				
NET_M	0.995	1.000			
NET_C	0.839	0.780	1.000		
SIZE	-0.019	-0.019	-0.013	1.000	
M_SIZE	0.625	0.619	0.551	0.092	1.000

This table reports the correlation between the degree centrality measures and the size measures in our dataset. The centrality measures of interest are degree centrality (NET), degree centrality computed using the managers' network only (NET_M) and degree centrality computed using the consultants' network only (NET_C). The centrality measures are computed across all asset classes in Panel A, in UK equities in Panel B, in UK bonds in Panel C and in international equities in Panel D. SIZE denotes the assets under management of each fund-manager pairing, while M_SIZE denotes each manager's assets under management across all funds managed. The size variables are converted to relative size by dividing them by the cross-sectional average and taking the natural log of this quantity. The centrality measures are computed to relative as well as the cross-section dimensions.

		Par	iel A. Ul	X Equitio	es			Pa	nel B. U.	K Bond				Pan	el C. Int	. Equition	se	
SIZE	-0.194 (0.14)	-0.200 (0.13)	-0.195 (0.14)	-0.202 (0.12)	-0.201 (0.13)	-0.200 (0.13)	-0.118 (0.05)	-0.105 (0.08)	-0.119 (0.04)	-0.106 (0.07)	-0.110 (0.06)	-0.109 (0.06)	-0.627 (0.00)	-0.629 (0.00)	-0.629 (0.00)	-0.628 (0.00)	-0.624 (0.00)	-0.631 (0.00)
M_SIZE	-0.336 (0.00)	-0.384 (0.00)	-0.323 (0.00)	-0.351 (0.00)	-0.325 (0.00)	-0.463 (0.00)	-0.026 (0.52)	$\begin{array}{c} 0.063 \\ (0.14) \end{array}$	-0.021 (0.59)	$\begin{array}{c} 0.045 \\ (0.25) \end{array}$	-0.083 (0.03)	-0.046 (0.39)	-0.158 (0.23)	-0.169 (0.20)	-0.140 (0.27)	-0.137 (0.28)	-0.183 (0.13)	-0.394 (0.02)
NET	$\begin{array}{c} 0.357 \\ (0.04) \end{array}$	$0.076 \\ (0.68)$					$\begin{array}{c} 0.184 \\ (0.06) \end{array}$	$\begin{array}{c} 0.301 \\ (0.00) \end{array}$					-0.104 (0.65)	-0.159 (0.57)				
$NET \times M_SIZE$		$0.326 \\ (0.00)$						-0.230 (0.00)						$\begin{array}{c} 0.051 \\ (0.69) \end{array}$				
$NET_{-}M$			$\begin{array}{c} 0.264 \\ (0.06) \end{array}$	$^{+0.008}_{-0.06}$					$\begin{array}{c} 0.137 \\ (0.05) \end{array}$	$\begin{array}{c} 0.240 \\ (0.00) \end{array}$					-0.165 (0.36)	-0.145 (0.51)		
$NET_{-}M \times M_{-}SIZE$				$\begin{array}{c} 0.317 \\ (0.00) \end{array}$						$-0.204 \\ (0.00)$						-0.019 (0.86)		
$NET_{-}C$					0.443 (0.03)	$0.622 \\ (0.01)$					$\begin{array}{c} 0.754 \\ (0.00) \end{array}$	$\begin{array}{c} 0.727 \\ (0.00) \end{array}$					$\begin{array}{c} 0.022 \\ (0.95) \end{array}$	$\begin{array}{c} 0.256 \\ (0.47) \end{array}$
$NET_C \times M_SIZE$						$0.254 \\ (0.01)$						-0.056 (0.32)						$\begin{array}{c} 0.337 \\ (0.02) \end{array}$
Between R^2	0.014	0.009	0.014	0.010	0.017	0.011	0.061	0.057	0.061	0.061	0.053	0.053	0.024	0.024	0.023	0.023	0.026	0.030
Joint-Significance	0.038	0.000	0.057	0.000	0.032	0.005	0.064	0.000	0.053	0.000	0.000	0.000	0.652	0.849	0.362	0.640	0.947	0.075

centrality. The risk-adjusted return of manager i in fund j at time t is computed as $\hat{r}_{i,j,t}^{adj} = \hat{\alpha}_{i,j} + \hat{\epsilon}_{i,j,t}$, where $\hat{\alpha}_{i,j}$ is estimated using the full set of observations available for manager i in fund This table reports results for panel regressions of fund-managers' risk-adjusted performance in UK equities (Panel A), UK bonds (Panel B) and International equities (Panel C) and managers' j. We adopt a four-factor model for UK Equities:

$$r_{i,j,t} = \alpha_{i,j} + \beta_{1,i,j} r_{m,t} + \beta_{2,i,j} SMB_t + \beta_{3,i,j} HML_t + \beta_{4,i,j} MOM_t + \epsilon_{i,j,t}$$

where $r_{m,t}$ is the excess return on the UK market portfolio, SMB_t is a size factor, HML_t is a value-growth factor and MOM_t is a momentum factor. We adopt a two-factor model for UK bonds:

$$r_{i,j,t} = \alpha_{i,j} + \beta_{1,i,j} GOVB_t + \beta_{2,i,j} CONS_t + \epsilon_{i,j,t},$$

where $GOVB_t$ is the excess return on the FTSE All-Gilts Total Return Index and $CONS_t$ is the excess return on the UK government consol bonds. Finally, we adopt a four-factor model for international equities:

$$r_{i,j,t} = \alpha_{i,j} + \beta_{1,i,j} \ NA_t + \beta_{2,i,j} \ EAFEX_t + \beta_{3,i,j} \ SMB_t + \beta_{4,i,j} \ HML_t + \epsilon_{i,j,t},$$

in UK equities (in Panel A), UK bonds (in Panel B) and international equities (in Panel C) across all funds managed (denoted by M_SIZE). The centrality measures of interest are degree observations. The panel regressions use as control variables the assets under management of each fund-manager pairing (denoted by SIZE), as well as each manager's assets under management where NA_t is the sterling-denominated excess return on the MSCI North American Total Return Index, $EAFEX_t$ is the sterling-denominated excess return on the MSCI Europe Australasia Far Eastern ex-UK Total Return Index, SMB_t is a global size factor and HML_t is a global value-growth factor. We drop from the sample the fund-manager pairings that have less than 12 centrality (NET), degree centrality computed using the managers' network only (NET-M) and degree centrality computed using the consultants' network only (NET-C), computed in UK natural log of this quantity. The centrality measures are converted to relative centrality by dividing them by the cross-sectional average. For each centrality measure we adopt two specifications. equities (Panel A), UK bonds (Panel B) or international equities (Panel C). The size variables are converted to relative size by dividing them by the cross-sectional average and taking the In the first we include the centrality measure as well as its square. In the second we include the centrality measure and its interaction with manager size. All the specifications use fund-manager and time fixed effects. The *p*-values are reported in parentheses and are computed using standard errors that are clustered at the fund-manager level.

Table 3. Centrality and Fund-Manager Performance

Table 4. Permutation tests for Consultants' Information Ratios in UK Equities

	$Consultant \ 1$	$Consultant \ 2$	Consultant 3	$Consultant \ 11$
Consultant 1	0	$0.008 \\ (0.81)$	$\begin{array}{c} 0.063 \\ (0.08) \end{array}$	$ \begin{array}{c} 0.186 \\ (0.00) \end{array} $
Consultant 2		0	$\begin{array}{c} 0.055 \\ (0.09) \end{array}$	$\begin{array}{c} 0.178 \\ (0.00) \end{array}$
Consultant 3			0	$ \begin{array}{c} 0.123 \\ (0.00) \end{array} $
Consultant 11				0

This table reports differences in average Information Ratios across the four consultants with the largest number of mandates, i.e. Consultant 1, Consultant 2, Consultant 3 and Consultant 11. The information ratio of consultant i in fund j is computed as

$$IR_{i,j} = \frac{\frac{1}{(t_{max} - t_{min} + 1)} \sum_{t_{min}}^{t_{max}} (r_{i,j,t} - r_{m,t})}{\sqrt{\frac{1}{(t_{max} - t_{min})} \sum_{t_{min}}^{t_{max}} \left((r_{i,j,t} - r_{m,t}) - \overline{(r_{i,j,\cdot} - r_{m,\cdot})} \right)^2}}$$

where t_{min} (t_{max}) is the first (last) observation of each consultant-fund pairing, $r_{i,j,t}$ is the excess return of consultant *i* in fund *j* at time *t* and $r_{m,t}$ is the excess return on the UK market portfolio at time *t*. We drop from the sample the consultant-manager pairings that have less than 12 observations. Each entry reports the difference average Information Ratio between the row consultant and the column consultant. In parentheses are reported the p-values computed using a permutation test with 10000 iterations. The information ratios are annualized by multiplying $IR_{i,j}$ by 2. The annualized average information ratio for consultant 1 is 0.114, for consultant 2 is 0.106, for consultant 3 is 0.051 and for consultant 4 is -0.072.

Centrality
Managers'
and
-Flows
Fund
Table 5.

	$\begin{array}{c} -0.006\\ (0.77)\\ (0.77)\\ -0.108\\ (0.00)\\ -0.456\\ (0.51)\end{array}\end{array}$	$\begin{array}{c} 0.014 \\ (0.52) \\ 0.020 \\ (0.01) \\ 0.981 \\ (0.00) \end{array}$	0.000	$\begin{array}{c} -0.016\\ -0.47\\ -0.47\\ 0.47\\ 0.140\\ 0.559\\ 0.559\\ 0.559\\ 0.559\\ 0.030\\ 0.550\\ 0.020\\ 0.030\\ 0.020\\ 0.020\\ 0.020\\ 0.063\\ \end{array}$	0.024
	$\begin{array}{c} -0.007\\ (0.73)\\ (0.096\\ (0.00)\\ -0.427\\ (0.54)\end{array}$	$\begin{array}{c} 0.008 \\ (0.70) \\ 0.965 \\ (0.00) \end{array}$	0.000	$\begin{array}{c} -0.017\\ (0.45)\\ -0.124\\ (0.00)\\ 0.423\\ (0.73)\\ 0.423\\ (0.73)\\ (0.73)\\ (0.73)\\ (0.73)\\ (0.73)\\ (0.57)\end{array}$	0.273
	$\begin{array}{c} -0.025\\ (0.25)\\ (0.107\\ -0.107\\ 0.235\\ 0.235\\ (0.73)\end{array}$	$\begin{array}{c} 0.044 \\ (0.03) \\ 0.019 \\ (0.01) \end{array}$	0.003	$\begin{array}{c} -0.017\\ -0.46)\\ -0.140\\ 0.700\\ 0.700\\ 0.700\\ 0.56)\\ 0.56)\\ 0.035\\ 0.035\\ 0.020\\ (0.02)\\ 0.020\end{array}$	0.022
	$\begin{array}{c} -0.025\\ (0.23)\\ (0.095\\ (0.00)\\ 0.251\\ 0.251\\ (0.71) \end{array}$	0.038 (0.06)	0.057	$\begin{array}{c} -0.018\\ -0.124\\ (0.43)\\ -0.124\\ (0.00)\\ 0.589\\ 0.589\\ (0.030\\ 0.589\\ (0.030\\ (0.22)\\ (0.22)\end{array}$	0.216
	$\begin{array}{c} -0.012\\ (0.58)\\ (0.099\\ -0.009\\ -0.002\\ (1.00)\\ (1.00)\\ (0.21)\\ 0.018\\ (0.21)\\ 0.018\\ (0.79)\\ 0.379\\ 0.379\\ 0.379\end{array}$	(00.0)	0.000	$\begin{array}{c} -0.018\\ (0.42)\\ -0.127\\ (0.00)\\ 0.777\\ (0.001)\\ 0.777\\ (0.51)\\ (0.51)\\ 0.001\\ (0.62)\\ (0.62)\\ (0.62)\end{array}$	0.184
uities	$\begin{array}{c} -0.012\\ (0.58)\\ -0.099\\ 0.009\\ 0.003\\ (1.00)\\ (1.00)\\ (0.12)\\ (0.377\\ 0.$	(00.0)	0.000	$\begin{array}{c} \text{onds} \\ \begin{array}{c} -0.018 \\ (0.42) \\ -0.126 \\ (0.00) \\ 0.778 \\ (0.51) \\ (0.51) \\ (0.51) \\ (0.61) \end{array} \end{array}$	0.160
r UK Eq	$\begin{array}{c} -0.024\\ (0.25)\\ (0.100\\ (0.100)\\ 0.243\\ (0.72)\\ (0.72)\\ (0.72)\\ (0.01)\\ (0.01)\\ (0.89) \end{array}$		0.003	br UK B. -0.018 -0.018 (0.43) 0.700 0.700 (0.55) (0.55) (0.14) (0.019 (0.14) (0.14) (0.001 (0.80)	0.116
esults fo	$\begin{array}{c} -0.024\\ (0.25)\\ (0.25)\\ (0.00)\\ 0.245\\ (0.71)\\ (0.71)\\ (0.73)\\ (0.00)\end{array}$		0.002	tesults for -0.018 -0.018 -0.018 -0.018 -0.126 -0.126 -0.126 -0.126 -0.126 -0.126 -0.001 -0.126 -0.001 -0.001 -0.000 -0.0	0.059
iel A. Ro	$\begin{array}{c} -0.010\\ (0.64)\\ (0.64)\\ (0.00)\\ (0.00)\\ (0.00)\\ (0.00)\\ (0.026)\\ (0.12)\\ (0.004)\\ (0.00)\\ (0.00)\\ (0.00) \end{array}$		0.000	mel B. F -0.017 -0.017 -0.130 -0.130 0.702 0.032 0.003 0.003 0.003 (0.15) (0.15) 0.003 0.003 (0.00) (0.90)	0.115
Paı	$\begin{array}{c} -0.010\\ (0.64)\\ (0.64)\\ (0.00)\\ -0.073\\ (0.91)\\ (0.09)\\ (0.09)\\ (0.09)\\ (0.09)\\ (0.00)\\ \end{array}$		0.000	P ² -0.017 -0.129 -0.129 (0.689 0.689 0.689 0.035 (0.08) (0.08)	0.119
	$\begin{array}{c} -0.024\\ (0.25)\\ (0.25)\\ (0.00)\\ 0.252\\ (0.71)\\ (0.71)\\ (0.71)\\ (0.043\\ 0.003\\ (0.03)\\ (0.63)\end{array}$		0.002	$\begin{array}{c} -0.017\\ (0.45)\\ -0.130\\ 0.677\\ 0.031\\ 0.003\\ 0.003\\ (0.66)\end{array}$	0.075
	$\begin{array}{c} -0.024 \\ (0.25) \\ (0.25) \\ 0.102 \\ (0.00) \\ 0.260 \\ (0.70) \\ (0.70) \\ (0.00) \end{array}$		0.002	$\begin{array}{c} -0.017\\ (0.45)\\ -0.129\\ (0.00)\\ 0.663\\ 0.058\\ (0.05)\\ (0.05) \end{array}$	0.046
	Lag-Flow M_SIZE Ret NET $NET \times M_SIZE$ $NET \times Ret$ NET M NET_M $NET_M \times M_SIZE$ $NET_M \times Ret$	NET.C $NET.C \times M.SIZE$ $NET.C \times Ret$	Joint-Significance	Lag-Flow M_SIZE Ret NET NET × M_SIZE NET × M_SIZE NET × Ret NET M × M_SIZE NET M × Ret NET_C NET_C × M_SIZE NET_C × Ret	Joint-Significance

			Panel C	C. Result	s for Int	ernation	al Equiti	es				
$Lag_{-}Flow$	-0.039	-0.039	-0.034	-0.035	-0.039	-0.039	-0.035	-0.035	-0.041	-0.039	-0.035	-0.033
M_SIZE	-0.109	-0.110	-0.109	-0.110	-0.107	-0.107	-0.107	-0.107	-0.102	-0.119	-0.102	-0.119
3et	(0.00) -0.207	-0.228	(0.00)	(0.00) -0.341	(0.00) -0.218	(0.00) -0.227	-0.296	-0.304	-0.191	-0.250	(0.00) -0.439	-0.486
VET	(0.03) (0.043)	(0.01) (0.034)	(0.48) (0.039)	(0.45) (0.030)	(0.03)	(0.62)	(26.0)	(06.0)	(7.9.0)	(86.0)	(0.34)	(62.0)
$VET \times M_SIZE$	(00.0)	(0.03)	(00.0)	(0.00)								
VET imes Ret		(02.0)	0.154	(0.21)								
V ETM			(60.0)	(20.0)	0.031	0.026	0.028	0.023				
$VETM \times MSIZE$					(00.0)	0.003	(00.0)	0.003				
${\sf VET}_{-M} imes Ret$						(0.40)	0.108	(0.40) (0.109)				
VETC							(0.04)	(0.03)	0.035	0.039	0.027	0.031
$VETC \times M_SIZE$									(U.U4)	(0.02) (0.025)	(21.0)	(0.00) (0.025)
$VETC \times Ret$										(00.0)	$\begin{array}{c} 0.318 \\ (0.01) \end{array}$	$\begin{array}{c}(0.00)\\0.304\\(0.01)\end{array}$
loint-Significance	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.042	0.000	0.006	0.000

(Table continued from the previous page)

This table reports panel regression results for the effect of managers' centrality, size and past performance on fund inflows and outflows. The fund-flow variable for manager i over quarter t is defined as:

$$Flow_{i,t:t+1} = \left(\frac{SMV_{i,t+1} - SMV_{i,t}}{SMV_{i,t}} - R_{i,t:t+1}\right) * SMV_{i,t}$$

return generated over quarter t + 1 in the respective asset class. The centrality measures of interest are degree centrality (NET), degree centrality computed using the managers' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-C), computed in UK equities (Panel B), UK bonds (Panel B) or international equities (Panel C). The size variable M_{-SIZE} denotes the total assets under management (across all asset classes and funds managed) of each manager at each point in time. We convert the size variable to relative size by dividing it by the cross-sectional average and taking the natural log of this quantity. The centrality measures are converted to relative centrality by dividing them by the cross-sectional where $SMV_{i,t}$ is the starting market value of manager i's asset holdings in UK equities (Panel A), UK bonds (Panel B) and international equities (Panel C) at quarter t and $R_{i,t:t+1}$ is the average. The past performance of each manager is constructed by value-weighting the returns in a given asset class across the various funds managed over the previous year. All results are computed using manager and time fixed effects. The *p*-values are reported in parentheses and are computed using standard errors that are clustered at the manager level.

											1							
		Pa.	nel A. U	K Equiti	ies			Ъ.	anel B. I	JK Bond	s			Рал	nel C. In	t. Equit	les	
SIZE	-1.291 (0.00)	-1.285 (0.00)	-1.292 (0.00)	-1.285 (0.00)	-1.289 (0.00)	-1.291 (0.00)	-0.507 (0.00)	-0.502 (0.00)	-0.501 (0.00)	-0.494 (0.00)	-0.504 (0.00)	-0.502 (0.00)	-1.704 (0.00)	-1.722 (0.00)	-1.703 (0.00)	-1.721 (0.00)	-1.717 (0.00)	-1.723 (0.00)
M_SIZE	-0.124 (0.09)	-0.072 (0.33)	-0.117 (0.10)	-0.090 (0.21)	-0.193 (0.00)	-0.014 (0.86)	-0.178 (0.00)	-0.139 (0.01)	-0.198 (0.00)	-0.160 (0.00)	-0.178 (0.00)	-0.141 (0.03)	-0.424 (0.00)	-0.490 (0.00)	-0.424 (0.00)	-0.469 (0.00)	-0.331 (0.00)	-0.519 (0.00)
NET	-0.102 (0.64)	$\begin{array}{c} 0.197 \\ (0.36) \end{array}$					-0.191 (0.10)	-0.139 (0.23)					$0.527 \\ (0.02)$	$\begin{array}{c} 0.180 \\ (0.47) \end{array}$				
$NET \times M_SIZE$		-0.347 (0.00)						-0.102 (0.05)						$\begin{array}{c} 0.321 \\ (0.00) \end{array}$				
NETM			-0.126 (0.48)	$\begin{array}{c} 0.126 \\ (0.48) \end{array}$					-0.067 (0.42)	$^{-0.007}_{(0.93)}$					$0.464 \\ (0.01)$	$\begin{array}{c} 0.160 \\ (0.42) \end{array}$		
$NET_M \times M_SIZE$				-0.293 (0.00)						-0.117 (0.01)						$\begin{array}{c} 0.292 \\ (0.00) \end{array}$		
NET_C					$\begin{array}{c} 0.365 \\ (0.04) \end{array}$	$\begin{array}{c} 0.135 \\ (0.51) \end{array}$					-0.281 (0.03)	-0.309 (0.03)					$\begin{array}{c} 0.148 \\ (0.58) \end{array}$	$0.356 \\ (0.22)$
$NET_C \times M_SIZE$						-0.327 (0.00)						-0.057 (0.34)						$\begin{array}{c} 0.301 \\ (0.02) \end{array}$
Between R^2	0.054	0.050	0.054	0.051	0.053	0.049	0.177	0.177	0.176	0.177	0.178	0.178	0.086	0.088	0.086	0.088	0.089	0.089
Joint-Significance	0.639	0.000	0.484	0.000	0.041	0.000	0.097	0.040	0.417	0.022	0.034	0.083	0.024	0.001	0.013	0.000	0.581	0.067
This table reports resul a four-factor model for	ts for pan UK Equit	el regress ies:	ions of fi	ınd-mane	ıgers' rish	¢ in UK eq	uities (Pau	nel A), U	K bonds	(Panel E	3) and In	cernational	equities (]	Panel C)	and man	lagers' ce	ntrality.	We adopt
where $r_{m,t}$ is the exces bonds:	s return o	n the UK	t market	$r_{i,j,t}$ portfolio	$\alpha = \alpha_{i,j} + SMB_t$	$-\beta_{1,i,j} r_m,$ is a size fa	$t + \beta_{2,i,j}$, ctor, HM .	$SMB_t + L_t$ is a vi	$eta_{3,i,j}\;Hl$ alue-grow	$ML_t + eta_4$ th factor	, <i>i</i> , <i>j</i> MOi and MC	$M_t + \epsilon_{i,j,t},$ M_t is a m	omentum	factor. V	Ve adopt	a two-fa	ctor mode	el for UK
where $GOVB_t$ is the epitemeteria equities:	ccess retur	n on the	FTSE A	ll-Gilts T	otal Retu	$r_{i,j,t} = \alpha_i$ 1rn Index i	$i,j + \beta_{1,i,j}$ and CON_{i}	$GOVB_t$. S_t is the	$+ \beta_{2,i,j} C$ excess ret	$ONS_t +$ turn on t	$\epsilon_{i,j,t},$ he UK go	overnment	consol bor	ıds. Fina	lly, we ac	lopt a foi	ur-factor 1	nodel for
where NA_t is the sterli Far Eastern ex-UK Tot	ng-denom al Return	inated ex Index, S	cess retu MB_t is i	$r_{i,j,t}$: rn on the ι global s	$= \alpha_{i,j} + \frac{1}{i}$, MSCI N ize factor	$\beta_{1,i,j} NA_t$ lorth Ame r and HM	$ + \beta_{2,i,j} E $ rican Tota L_t is a glc	$AFEX_t$ $AFEX_t$ $AFEX_t$	$+ \beta_{3,i,j} $ Index, E e-growth	$SMB_t + AFEX_t$ factor. V	$\beta_{4,i,j} H h$ is the ste Ve drop f	$AL_t + \epsilon_{i,j,i}$ rrling-denoi rom the sa	t, minated ex mple the f	ccess retu îund-man	ırn on th ager pain	e MSCI F rings that	Europe Aı have les	ıstralasia s than 12
obcommetions The walt	Jon South		1 + 0 · 7 + +	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	104.000 C	() o o o o	<u>()</u>	Tho are	mon long	in paoiooo		ldoinon lond		oto un dor	0.0000000	fo toom	bard door	

Table 6. Centrality and Fund-Manager Risk

48

observations. The risk of manager *i* in fund *j* at time *t* is computed as $\widehat{risk}_{i,j,t} = [\widehat{e}_{i,j,t}]$. The panel regressions use as control variables the assets under management of each fund-manager pairing (denoted by SIZE), as well as each manager's assets under management in UK equities (in Panel A), UK bonds (in Panel B) and international equities (in Panel C) across all funds cross-sectional average. For each centrality measure we adopt two specifications. In the first we include the centrality measure as well as its square. In the second we include the centrality measure and its interaction with manager size. All the specifications use fund-manager and time fixed effects. The *p*-values are reported in parentheses and are computed using standard errors managed (denoted by M_-SIZE). The centrality measures of interest are degree centrality (NET), degree centrality computed using the managers' network only (NET-M) and degree centrality computed using the consultants' network only (NET-C), computed in UK equities (Panel A), UK bonds (Panel B) or international equities (Panel C). The size variables are converted to relative size by dividing them by the cross-sectional average and taking the natural log of this quantity. The centrality measures are converted to relative centrality by dividing them by the that are clustered at the fund-manager level. Fa:

	U	K Equit	ies	τ	JK Bond	ls	In	t. Equit	ies
$Risk_Adj_Ret$	-0.016 (0.00)	-0.016 (0.00)	-0.017 (0.00)	-0.004 (0.50)	-0.004 (0.51)	-0.004 (0.46)	-0.009 (0.00)	-0.009 (0.00)	-0.010 (0.00)
SIZE	-0.131 (0.00)	-0.135 (0.00)	-0.122 (0.00)	-0.126 (0.00)	-0.127 (0.00)	-0.123 (0.00)	-0.160 (0.00)	-0.164 (0.00)	-0.153 (0.00)
NET	-0.627 (0.00)			-0.063 (0.04)			-0.599 (0.00)		
NET_M		-0.537 (0.00)			-0.041 (0.09)			-0.513 (0.00)	
NET_C			-0.911 (0.00)			-0.149 (0.00)			-0.825 (0.00)

Table 7. Survival Analysis for Managers and Consultants

Panel A. Analysis at the Managers' Level

Panel B. Analysis at the Consultants' Level

	UK Equities	UK Bonds	Int. Equities
Risk_Adj_Ret	0.001	-0.010	-0.008
	(0.89)	(0.14)	(0.00)
SIZE	-0.205	-0.172	-0.237
	(0.00)	(0.00)	(0.00)
NET	-0.696	-0.463	-0.522
	(0.00)	(0.00)	(0.00)

This table reports in Panel A (Panel B) the coefficients of a Cox proportional hazard rate model relating the probability of managers' (consultants') contracts being terminated in UK equities, UK bonds and international equities asset classes to their past performance, their size, as well as their network centrality. In Panel A, SIZE denotes the assets under management of each fund-manager pairing. In Panel B, SIZE denotes the assets under management of each fund-manager pairing. In Panel B, SIZE denotes the assets under management of each fund-consultant pairing. Past performance ($Risk_Adj_Ret$) is computed as the average abnormal returns in UK equities, UK bonds or international equities over the previous two quarters for each fund-manager pairing in Panel A and for each fund-consultant pairing in Panel B. In Panel A the centrality measures of interest are managers' degree centrality (NET), managers' degree centrality computed using the managers' network only (NET_M) and managers' degree centrality computed using the consultants' network only (NET_C), computed in UK equities, UK bonds or international equities. In Panel B the centrality measure of interest are consultants' degree centrality (NET). The size variables are converted to relative size by dividing them by the cross-sectional average and taking the natural log of this quantity. The centrality measures are converted to relative centrality by dividing them by the cross-sectional average. The *p*-values are reported in parentheses.

Table 8. Granger Causality Tests of Size versus Network CentralityPanel A. Dependent Variable: Centrality Measure

				I								
	A.I. All	l Asset (Classes	A.II.	UK Eqt	iities	A.III	. UK Be	spuc	A.IV.	Int. Eq	lities
M_SIZE	-0.001 (0.39)	-0.001 (0.39)	-0.000 (0.01)	0.001 (0.12)	$\begin{array}{c} 0.001 \\ (0.22) \end{array}$	$\begin{array}{c} 0.000\\ (0.36) \end{array}$	0.000 (0.48)	$0.000 \\ (0.79)$	0.000 (0.33)	-0.000 (0.44)	-0.001 (0.29)	$\begin{array}{c} 0.000\\ (0.28) \end{array}$
NET	0.885 (0.00)			$\begin{array}{c} 0.865 \\ (0.00) \end{array}$			$\begin{array}{c} 0.874 \\ (0.00) \end{array}$			$0.890 \\ (00.0)$		
$NET_{-}M$		(00.0)			(0.00)			(00.0)			$\begin{array}{c} 0.889 \\ (0.00) \end{array}$	
NETC			$\begin{array}{c} 0.840 \\ (0.00) \end{array}$			0.838 (0.00)			$0.742 \\ (0.00)$			$\begin{array}{c} 0.834 \\ (0.00) \end{array}$

Panel B. Dependent Variable: Size

	B.I. Al	ll Asset (Classes	B.II.	UK Eqı	iities	B.II.	I. UK B	onds	B.IV.	Int. Eq	uities
M_SIZE	0.619 (0.00)	0.622 (0.00)	$0.650 \\ (0.00)$	0.829 (0.00)	$\begin{array}{c} 0.836\\ (0.00) \end{array}$	0.838 (0.00)	0.798 (00.0)	(00.0) (00.0)	0.856 (0.00)	0.780 (0.00)	0.783 (0.00)	$\begin{array}{c} 0.791 \\ (0.00) \end{array}$
NET	$2.321 \\ (0.00)$			$1.278 \\ (0.01)$			$2.221 \\ (0.00)$			$\begin{array}{c} 0.973 \\ (0.02) \end{array}$		
$NET_{-}M$		$2.140 \\ (0.00)$			$\begin{array}{c} 0.915 \\ (0.04) \end{array}$			2.086 (0.00)			$\begin{array}{c} 0.858\\ (0.07) \end{array}$	
NETC			$7.376 \\ (0.04)$			$3.510 \\ (0.02)$			$1.475 \\ (0.33)$			1.438 (0.42)

This table reports the results of panel Granger causality tests for managers' size and centrality. Managers' centrality is computed as the degree centrality (NET), degree centrality computed using the managers' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network only (NET-M) and degree centrality computed using the consultants' network on the consultant s' network on the consultant s' net management across funds and asset classes managed. The dependent variable is one of the centrality measures in Panel A and managers' size in Panel B. Centrality measures are computed across asset classes in Panels A.I. in UK equities in Panels A.II. and B.II., in UK bonds in Panels A.III. and B.III. and in international equities in Panels A.IV. and B.IV. The parameters are estimated using one-step GMM that use up to 16 lags of the dependent variable as instruments. The details of the procedure are reported in Section 6 of the paper. The *p*-values are reported in parentheses and are computed using robust standard errors. CFR working paper series

CFR Working Papers are available for download from www.cfr-cologne.de.

Hardcopies can be ordered from: Centre for Financial Research (CFR), Albertus Magnus Platz, 50923 Koeln, Germany.

No.	Author(s)	Title
15-16	D. Blake, A.G. Rossi, A. Timmermann, I. Tonks, R. Wermers	Network Centrality and Pension Fund Performance
15-15	S. Jank, E. Smajbegovic	Dissecting Short-Sale Performance: Evidence from Large Position Disclosures
15-14	M. Doumet, P. Limbach, E. Theissen	Ich bin dann mal weg: Werteffekte von Delistings deutscher Aktiengesellschaften nach dem Frosta-Urteil
15-13	G. Borisova, P.K. Yadav	Government Ownership, Informed Trading and Private Information
15-12	V. Agarwal, G.O. Aragon, Z. Shi	Funding Liquidity Risk of Funds of Hedge Funds: Evidence from their Holdings
15-11	L. Ederington, W. Guan, P.K. Yadav	Dealer spreads in the corporate Bond Market: Agent vs. Market-Making Roles
15-10	J.R. Black, D. Stock, P.K. Yadav	The Pricing of Different Dimensions of Liquidity: Evidence from Government Guaranteed Bank Bonds
15-09	V. Agarwal, H. Zhao	Interfund lending in mutual fund families: Role of internal capital markets
15-08	V. Agarwal, T. C. Green, H. Ren	Alpha or Beta in the Eye of the Beholder: What drives Hedge Fund Flows?
15-07	V. Agarwal, S. Ruenzi, F. Weigert	Tail risk in hedge funds: A unique view from portfolio holdings
15-06	C. Lan, F. Moneta, R. Wermers	Mutual Fund Investment Horizon and Performance
15-05	L.K. Dahm, C. Sorhage	Milk or Wine: Mutual Funds' (Dis)economies of Life
15-04	A. Kempf, D. Mayston, M. Gehde-Trapp, P. K. Yadav	Resiliency: A Dynamic View of Liquidity
15-03	V. Agarwal, Y. E. Arisoy, N. Y. Naik	Volatility of Aggregate Volatility and Hedge Funds Returns

15-02	G. Cici, S. Jaspersen, A. Kempf	Speed of Information Diffusion within Fund Families
15-01	M. Baltzer, S. Jank, E. Smajlbegovic	Who trades on momentum?

No.	Author(s)	Title
14-14	G. Cici, L. K. Dahm, A. Kempf	Trading Efficiency of Fund Families: Impact on Fund Performance and Investment Behavior
14-13	V. Agarwal, Y. Lu, S. Ray	Under one roof: A study of simultaneously managed hedge funds and funds of hedge funds
14-12	P. Limbach, F. Sonnenburg	Does CEO Fitness Matter?
14-11	G. Cici, M. Gehde-Trapp, M. Göricke, A. Kempf	What They Did in their Previous Life: The Investment Value of Mutual Fund Managers' Experience outside the Financial Sector
14-10	O. Korn, P. Krischak, E. Theissen	Illiquidity Transmission from Spot to Futures Markets
14-09	E. Theissen, L. S. Zehnder	Estimation of Trading Costs: Trade Indicator Models Revisited
14-08	C. Fink, E. Theissen	Dividend Taxation and DAX Futures Prices
14-07	F. Brinkmann, O. Korn	Risk-adjusted Option-implied Moments
14-06	J. Grammig, J. Sönksen	Consumption-Based Asset Pricing with Rare Disaster Risk
14-05	J. Grammig, E. Schaub	Give me strong moments and time – Combining GMM and SMM to estimate long-run risk asset pricing
14-04	C. Sorhage	Outsourcing of Mutual Funds' Non-core Competencies
14-03	D. Hess, P. Immenkötter	How Much Is Too Much? Debt Capacity And Financial Flexibility
14-02	C. Andres, M. Doumet, E. Fernau, E. Theissen	The Lintner model revisited: Dividends versus total payouts
14-01	N.F. Carline, S. C. Linn, P. K. Yadav	Corporate Governance and the Nature of Takeover Resistance

No.	Author(s)	Title
13-11	R. Baule, O. Korn, S. Saßning	Which Beta is Best? On the Information Content of Option-implied Betas
13-10	V. Agarwal, L. Ma, K. Mullally	Managerial Multitasking in the Mutual Fund Industry
13-09	M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers	Seasonal Asset Allocation: Evidence from Mutual Fund Flows

13-08	F. Brinkmann, A. Kempf, O. Korn	Forward-Looking Measures of Higher-Order Dependencies with an Application to Portfolio Selection
13-07	G. Cici, S. Gibson, Y. Gunduz, J.J. Merrick, Jr.	Market Transparency and the Marking Precision of Bond Mutual Fund Managers
13-06	S. Bethke, M. Gehde- Trapp, A. Kempf	Investor Sentiment, Flight-to-Quality, and Corporate Bond Comovement
13-05	P. Schuster, M. Trapp, M. Uhrig-Homburg	A Heterogeneous Agents Equilibrium Model for the Term Structure of Bond Market Liquidity
13-04	V. Agarwal, K. Mullally, Y. Tang, B. Yang	Mandatory Portfolio Disclosure, Stock Liquidity, and Mutual Fund Performance
13-03	V. Agarwal, V. Nanda, S.Ray	Institutional Investment and Intermediation in the Hedge Fund Industry
13-02	C. Andres, A. Betzer, M. Doumet, E. Theissen	Open Market Share Repurchases in Germany: A Conditional Event Study Approach
13-01	J. Gaul, E. Theissen	A Partially Linear Approach to Modelling the Dynamics of Spot and Futures Price

No.	Author(s)	Title
12-12	M. Gehde-Trapp, Y. Gündüz, J. Nasev	The liquidity premium in CDS transaction prices: Do frictions matter?
12-11	Y. Wu, R. Wermers, J. Zechner	Governance and Shareholder Value in Delegated Portfolio Management: The Case of Closed-End Funds
12-10	M. Trapp, C. Wewel	Transatlantic Systemic Risk
12-09	G. Cici, A. Kempf, C. Sorhage	Do Financial Advisors Provide Tangible Benefits for Investors? Evidence from Tax-Motivated Mutual Fund Flows
12-08	S. Jank	Changes in the composition of publicly traded firms: Implications for the dividend-price ratio and return predictability
12-07	G. Cici, C. Rosenfeld	The Investment Abilities of Mutual Fund Buy-Side Analysts
12-06	A. Kempf, A. Pütz, F. Sonnenburg	Fund Manager Duality: Impact on Performance and Investment Behavior
12-05	R. Wermers	Runs on Money Market Mutual Funds
12-04	R. Wermers	A matter of style: The causes and consequences of style drift in institutional portfolios
12-02	C. Andres, E. Fernau, E. Theissen	Should I Stay or Should I Go? Former CEOs as Monitors
12-01	L. Andreu, A. Pütz	Choosing two business degrees versus choosing one: What does it tell about mutual fund managers' investment behavior?

No.	Author(s)	Title
11-16	V. Agarwal, JP. Gómez, R. Priestley	Management Compensation and Market Timing under Portfolio Constraints
11-15	T. Dimpfl, S. Jank	Can Internet Search Queries Help to Predict Stock Market Volatility?
11-14	P. Gomber, U. Schweickert, E. Theissen	Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach
11-13	D. Hess, S. Orbe	Irrationality or Efficiency of Macroeconomic Survey Forecasts? Implications from the Anchoring Bias Test
11-12	D. Hess, P. Immenkötter	Optimal Leverage, its Benefits, and the Business Cycle
11-11	N. Heinrichs, D. Hess, C. Homburg, M. Lorenz, S. Sievers	Extended Dividend, Cash Flow and Residual Income Valuation Models – Accounting for Deviations from Ideal Conditions
11-10	A. Kempf, O. Korn, S. Saßning	Portfolio Optimization using Forward - Looking Information
11-09	V. Agarwal, S. Ray	Determinants and Implications of Fee Changes in the Hedge Fund Industry
11-08	G. Cici, LF. Palacios	On the Use of Options by Mutual Funds: Do They Know What They Are Doing?
11-07	V. Agarwal, G. D. Gay, L. Ling	Performance inconsistency in mutual funds: An investigation of window-dressing behavior
11-06	N. Hautsch, D. Hess, D. Veredas	The Impact of Macroeconomic News on Quote Adjustments, Noise, and Informational Volatility
11-05	G. Cici	The Prevalence of the Disposition Effect in Mutual Funds' Trades
11-04	S. Jank	Mutual Fund Flows, Expected Returns and the Real Economy
11-03	G.Fellner, E.Theissen	Short Sale Constraints, Divergence of Opinion and Asset Value: Evidence from the Laboratory
11-02	S.Jank	Are There Disadvantaged Clienteles in Mutual Funds?
11-01	V. Agarwal, C. Meneghetti	The Role of Hedge Funds as Primary Lenders

No.	Author(s)	Title
10-20	G. Cici, S. Gibson, J.J. Merrick Jr.	Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds
10-19	J. Hengelbrock, E. Theissen, C. Westheide	Market Response to Investor Sentiment
10-18	G. Cici, S. Gibson	The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings
10-17	D. Hess, D. Kreutzmann, O. Pucker	Projected Earnings Accuracy and the Profitability of Stock Recommendations

10-16	S. Jank, M. Wedow	Sturm und Drang in Money Market Funds: When Money Market Funds Cease to Be Narrow
10-15	G. Cici, A. Kempf, A. Puetz	The Valuation of Hedge Funds' Equity Positions
10-14	J. Grammig, S. Jank	Creative Destruction and Asset Prices
10-13	S. Jank, M. Wedow	Purchase and Redemption Decisions of Mutual Fund Investors and the Role of Fund Families
10-12	S. Artmann, P. Finter, A. Kempf, S. Koch, E. Theissen	The Cross-Section of German Stock Returns: New Data and New Evidence
10-11	M. Chesney, A. Kempf	The Value of Tradeability
10-10	S. Frey, P. Herbst	The Influence of Buy-side Analysts on Mutual Fund Trading
10-09	V. Agarwal, W. Jiang, Y. Tang, B. Yang	Uncovering Hedge Fund Skill from the Portfolio Holdings They Hide
10-08	V. Agarwal, V. Fos, W. Jiang	Inferring Reporting Biases in Hedge Fund Databases from Hedge Fund Equity Holdings
10-07	V. Agarwal, G. Bakshi, J. Huij	Do Higher-Moment Equity Risks Explain Hedge Fund Returns?
10-06	J. Grammig, F. J. Peter	Tell-Tale Tails: A data driven approach to estimate unique market information shares
10-05	K. Drachter, A. Kempf	Höhe, Struktur und Determinanten der Managervergütung- Eine Analyse der Fondsbranche in Deutschland
10-04	J. Fang, A. Kempf, M. Trapp	Fund Manager Allocation
10-03	P. Finter, A. Niessen- Ruenzi, S. Ruenzi	The Impact of Investor Sentiment on the German Stock Market
10-02	D. Hunter, E. Kandel, S. Kandel, R. Wermers	Mutual Fund Performance Evaluation with Active Peer Benchmarks
10-01	S. Artmann, P. Finter, A. Kempf	Determinants of Expected Stock Returns: Large Sample Evidence from the German Market

No.	Author(s)	Title
09-17	E. Theissen	Price Discovery in Spot and Futures Markets: A Reconsideration
09-16	М. Тгарр	Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-15	A. Betzer, J. Gider, D.Metzger, E. Theissen	Strategic Trading and Trade Reporting by Corporate Insiders
09-14	A. Kempf, O. Korn, M. Uhrig-Homburg	The Term Structure of Illiquidity Premia
09-13	W. Bühler, M. Trapp	Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets
09-12	W. Bühler, M. Trapp	Explaining the Bond-CDS Basis – The Role of Credit Risk and Liquidity

09-11	S. J. Taylor, P. K. Yadav, Y. Zhang	Cross-sectional analysis of risk-neutral skewness
09-10	A. Kempf, C. Merkle, A. Niessen-Ruenzi	Low Risk and High Return – Affective Attitudes and Stock Market Expectations
09-09	V. Fotak, V. Raman, P. K. Yadav	Naked Short Selling: The Emperor's New Clothes?
09-08	F. Bardong, S.M. Bartram, P.K. Yadav	Informed Trading, Information Asymmetry and Pricing of Information Risk: Empirical Evidence from the NYSE
09-07	S. J. Taylor , P. K. Yadav, Y. Zhang	The information content of implied volatilities and model-free volatility expectations: Evidence from options written on individual stocks
09-06	S. Frey, P. Sandas	The Impact of Iceberg Orders in Limit Order Books
09-05	H. Beltran-Lopez, P. Giot, J. Grammig	Commonalities in the Order Book
09-04	J. Fang, S. Ruenzi	Rapid Trading bei deutschen Aktienfonds: Evidenz aus einer großen deutschen Fondsgesellschaft
09-03	A. Banegas, B. Gillen, A. Timmermann, R. Wermers	The Cross-Section of Conditional Mutual Fund Performance in European Stock Markets
09-02	J. Grammig, A. Schrimpf, M. Schuppli	Long-Horizon Consumption Risk and the Cross-Section of Returns: New Tests and International Evidence
09-01	O. Korn, P. Koziol	The Term Structure of Currency Hedge Ratios

No.	Author(s)	Title
08-12	U. Bonenkamp, C. Homburg, A. Kempf	Fundamental Information in Technical Trading Strategies
08-11	O. Korn	Risk Management with Default-risky Forwards
08-10	J. Grammig, F.J. Peter	International Price Discovery in the Presence of Market Microstructure Effects
08-09	C. M. Kuhnen, A. Niessen	Public Opinion and Executive Compensation
08-08	A. Pütz, S. Ruenzi	Overconfidence among Professional Investors: Evidence from Mutual Fund Managers
08-07	P. Osthoff	What matters to SRI investors?
08-06	A. Betzer, E. Theissen	Sooner Or Later: Delays in Trade Reporting by Corporate Insiders
08-05	P. Linge, E. Theissen	Determinanten der Aktionärspräsenz auf Hauptversammlungen deutscher Aktiengesellschaften
08-04	N. Hautsch, D. Hess, C. Müller	Price Adjustment to News with Uncertain Precision
08-03	D. Hess, H. Huang, A. Niessen	How Do Commodity Futures Respond to Macroeconomic News?
08-02	R. Chakrabarti, W. Megginson, P. Yadav	Corporate Governance in India

Setting a Fox to Keep the Geese - Does the Comply-or-Explain Principle Work?

2007

No.	Author(s)	Title
07-16	M. Bär, A. Niessen, S. Ruenzi	The Impact of Work Group Diversity on Performance: Large Sample Evidence from the Mutual Fund Industry
07-15	A. Niessen, S. Ruenzi	Political Connectedness and Firm Performance: Evidence From Germany
07-14	O. Korn	Hedging Price Risk when Payment Dates are Uncertain
07-13	A.Kempf, P. Osthoff	SRI Funds: Nomen est Omen
07-12	J. Grammig, E. Theissen, O. Wuensche	Time and Price Impact of a Trade: A Structural Approach
07-11	V. Agarwal, J. R. Kale	On the Relative Performance of Multi-Strategy and Funds of Hedge Funds
07-10	M. Kasch-Haroutounian, E. Theissen	Competition Between Exchanges: Euronext versus Xetra
07-09	V. Agarwal, N. D. Daniel, N. Y. Naik	Do hedge funds manage their reported returns?
07-08	N. C. Brown, K. D. Wei, R. Wermers	Analyst Recommendations, Mutual Fund Herding, and Overreaction in Stock Prices
07-07	A. Betzer, E. Theissen	Insider Trading and Corporate Governance: The Case of Germany
07-06	V. Agarwal, L. Wang	Transaction Costs and Value Premium
07-05	J. Grammig, A. Schrimpf	Asset Pricing with a Reference Level of Consumption: New Evidence from the Cross-Section of Stock Returns
07-04	V. Agarwal, N.M. Boyson, N.Y. Naik	Hedge Funds for retail investors? An examination of hedged mutual funds
07-03	D. Hess, A. Niessen	The Early News Catches the Attention: On the Relative Price Impact of Similar Economic Indicators
07-02	A. Kempf, S. Ruenzi, T. Thiele	Employment Risk, Compensation Incentives and Managerial Risk Taking - Evidence from the Mutual Fund Industry -
07-01	M. Hagemeister, A. Kempf	CAPM und erwartete Renditen: Eine Untersuchung auf Basis der Erwartung von Marktteilnehmern

No.	Author(s)	Title
06-13	S. Čeljo-Hörhager, A. Niessen	How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study
06-12	R. Wermers, Y. Wu, J. Zechner	Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers
06-11	U. v. Lilienfeld-Toal, S. Ruenzi	Why Managers Hold Shares of Their Firm: An Empirical Analysis
06-10	A. Kempf, P. Osthoff	The Effect of Socially Responsible Investing on Portfolio Performance

06-09	R. Wermers, T. Yao, J. Zhao	Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach
06-08	M. Hoffmann, B. Kempa	The Poole Analysis in the New Open Economy Macroeconomic Framework
06-07	K. Drachter, A. Kempf, M. Wagner	Decision Processes in German Mutual Fund Companies: Evidence from a Telephone Survey
06-06	J.P. Krahnen, F.A. Schmid, E. Theissen	Investment Performance and Market Share: A Study of the German Mutual Fund Industry
06-05	S. Ber, S. Ruenzi	On the Usability of Synthetic Measures of Mutual Fund Net-Flows
06-04	A. Kempf, D. Mayston	Liquidity Commonality Beyond Best Prices
06-03	O. Korn, C. Koziol	Bond Portfolio Optimization: A Risk-Return Approach
06-02	O. Scaillet, L. Barras, R. Wermers	False Discoveries in Mutual Fund Performance: Measuring Luck in Estimated Alphas
06-01	A. Niessen, S. Ruenzi	Sex Matters: Gender Differences in a Professional Setting

No.	Author(s)	Title
05-16	E. Theissen	An Analysis of Private Investors' Stock Market Return Forecasts
05-15	T. Foucault, S. Moinas, E. Theissen	Does Anonymity Matter in Electronic Limit Order Markets
05-14	R. Kosowski, A. Timmermann, R. Wermers, H. White	Can Mutual Fund "Stars" Really Pick Stocks? New Evidence from a Bootstrap Analysis
05-13	D. Avramov, R. Wermers	Investing in Mutual Funds when Returns are Predictable
05-12	K. Griese, A. Kempf	Liquiditätsdynamik am deutschen Aktienmarkt
05-11	S. Ber, A. Kempf, S. Ruenzi	Determinanten der Mittelzuflüsse bei deutschen Aktienfonds
05-10	M. Bär, A. Kempf, S. Ruenzi	Is a Team Different From the Sum of Its Parts? Evidence from Mutual Fund Managers
05-09	M. Hoffmann	Saving, Investment and the Net Foreign Asset Position
05-08	S. Ruenzi	Mutual Fund Growth in Standard and Specialist Market Segments
05-07	A. Kempf, S. Ruenzi	Status Quo Bias and the Number of Alternatives - An Empirical Illustration from the Mutual Fund Industry
05-06	J. Grammig, E. Theissen	Is Best Really Better? Internalization of Orders in an Open Limit Order Book
05-05	H. Beltran-Lopez, J. Grammig, A.J. Menkveld	Limit order books and trade informativeness
05-04	M. Hoffmann	Compensating Wages under different Exchange rate Regimes
05-03	M. Hoffmann	Fixed versus Flexible Exchange Rates: Evidence from Developing Countries
05-02	A. Kempf, C. Memmel	Estimating the Global Minimum Variance Portfolio
05-01	S. Frey, J. Grammig	Liquidity supply and adverse selection in a pure limit order book market

No.	Author(s)	Title
04-10	N. Hautsch, D. Hess	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery
04-09	A. Kempf, K. Kreuzberg	Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation
04-08	N.F. Carline, S.C. Linn, P.K. Yadav	Operating performance changes associated with corporate mergers and the role of corporate governance
04-07	J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav	Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze
04-06	N.Y. Naik, P.K. Yadav	Trading Costs of Public Investors with Obligatory and Voluntary Market-Making: Evidence from Market Reforms
04-05	A. Kempf, S. Ruenzi	Family Matters: Rankings Within Fund Families and Fund Inflows
04-04	V. Agarwal, N.D. Daniel, N.Y. Naik	Role of Managerial Incentives and Discretion in Hedge Fund Performance
04-03	V. Agarwal, W.H. Fung, J.C. Loon, N.Y. Naik	Risk and Return in Convertible Arbitrage: Evidence from the Convertible Bond Market
04-02	A. Kempf, S. Ruenzi	Tournaments in Mutual Fund Families
04-01	I. Chowdhury, M. Hoffmann, A. Schabert	Inflation Dynamics and the Cost Channel of Monetary Transmission

centre for Financial Research

cfr/university of cologne Albertus-Magnus-Platz D-50923 cologne Fon +49[0]221-470-6995 Fax +49[0]221-470-3992 Kempf@cfr-cologne.de WWW.cfr-cologne.de